Mailed

DEC 2 4 1991

Decision 91-12-053 December 18, 1991

BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

In the Matter of the Application of DEL ESTE WATER COMPANY (U 175 W), a corporation, for an order authorizing it to execute a Memorandum of Understanding with Modesto Irrigation District and the City of Modesto providing for a treated water supply.

Application 91-05-025 (Filed May 9, 1991)

OPINION

1. Summary of Decision

In this decision the Commission authorizes applicant Del Este Water Company (Del Este) to execute a certain Memorandum of Understanding (MOU) among Del Este, the Modesto Irrigation District (District), and the City of Modesto (City), providing for the construction of a project (the Project) which will enable the District to furnish surface water to the other parties. The Project will relieve Del Este from its dependency upon a dwindling and deteriorating supply of groundwater, presently Del Este's only water source. The Commission also finds that Del Este's execution of the MOU is reasonable and prudent, and in the best interests of its customers. This is a final order, and the proceeding is accordingly closed.

2. Background and Description of the Application

2.1 General .

Del Este seeks the Commission's authorization to execute the MOU, which essentially provides for construction of the Project to enable the District to deliver 30 million gallons (MG) of surface water per day initially, and up to 60 MG per day

ultimately, to the City and Del Este. 1 Entering into the MOU will free Del Este from its total reliance upon its system of wells, from which its entire water supply now comes, and which has suffered from steadily declining water tables and degradation of water quality in recent years.

Del Este states that it needs the treated water which the Project will supply, and anticipates that this water will constitute approximately 50% of the total supply furnished to its customers in the affected service area during summer months, and 100% during the winter months. With certain stipulated conditions, the Commission's Water Utilities Branch (Branch) recommends approval of the Application. In recognition of Del Este's customers' need for a continuing supply of water which meets increasingly stringent water quality standards, and the declining availability of such water from Del Este's existing underground sources, we agree that the approval sought by Del Este is required. We therefore grant the Application on the terms and conditions set forth in the Order.

on the second of the control of the second o

. ಮಂದು ಮಾರಣ ಮುಂದು ಕಾರಣದ ಮಾಡು ಕಾರಣದ ಕಾರ್ಮ

and the second of the second o

i destruire en la completa la completa de la completa del completa de la completa de la completa del completa de la completa de la completa de la completa de la completa del completa del completa de la completa de la completa del completa de la completa del com

¹ Del Este's receipt of California Public Utilities Commission approval to execute the MOU is an express condition precedent to Del Este's entering into the agreement. (Appendix A, p. 50.) Del Este states only that its Application is made "pursuant to the provisions of the Public Utilities Code...and General Order No. 96-A." (Application, p. 22.) Although the latter order does not, on its face, require Del Este to seek, nor authorize the Commission to give, such approval, we do so pursuant to Section 701 of the California Public Utilities Code, which states:

[&]quot;The commission may supervise and regulate every public utility in the State and may do all things, whether specifically designated in this part or in addition thereto, which are necessary and convenient in the exercise of such power and jurisdiction."

2.2 Procedural History and a second of the second cyledomia are

This Application is an outgrowth of Del Este's request for a general rate increase in Application (A.) 91-03-024. In that proceeding Del Este made a separate motion, specifically seeking the Commission's authority to enter into the MOU. At the behest of the Branch, Del Este withdrew its motion and refiled its request in the form of the instant Application. See Decision (D.) 91-12-073 in A.91-03-024, pp. 2-3 (mimeo.).

Branch prepared and served upon interested parties a report on the MOU dated October 1991, stating that Branch was concerned about the relative priority to be accorded Dol Este's customers under the MOU, and about the possibility that they could experience adversely discriminatory treatment because their interests might not be represented adequately on the District's Board of Directors. The report also makes certain recommendations to address these concerns. By written stipulation executed November 13, 1991, Del Este and Branch agreed, inter alia, that the sections of the report setting forth these concerns and recommendations be deemed to have been deleted. As part of the stipulation Branch also recommends that the Commission grant Del Este's Application and find that execution of the MOU by Del Este is reasonable and prudent and in the best interests of its customers.

2.3 Background of the MOU

According to the Application, most of Del Este's system (and approximately 80% of its customers) are within the District's boundaries. All of Del Este's supply for that portion of its system is currently drawn from underground. A water management study dated November 1984 (the Montgomery Study), which was prepared for the District and the City by James M. Montgomery Consulting Engineers, Inc., found that Del Este's and the City's groundwater sources were limited, supplies were declining, and a continued decline in the quantity of groundwater would result in a

gradual degradation of its quality. This fact, in combination with the prospect of increasingly stringent federal and state drinking water standards, portended growing difficulty in meeting domestic water supply needs solely through the use of groundwater. The Montgomery Study therefore recommended that the City and Del Este contract with the District for treated surface water to supplement their groundwater supplies.

According to the Application, the District was formed more than 100 years ago for the purpose of providing irrigation water to serve the needs of agriculture. To that end, the District perfected water rights sufficient to meet those needs. However, in recent years population growth within the District's boundaries has reduced the irrigated acreage while increasing the demand for municipal uses of water. The portions of the service areas of the City and Del Este which would be served with water pursuant to the MOU are within the boundaries of the District. By furnishing this water under an agreement, the District would simply shift the use of its water from agricultural to domestic purposes.

Based upon these circumstances, Del Este and the City concluded that it would be in their best interests and those of their respective customers to utilize the District's available water. The District has sufficient water rights in the Tuolomne River to enable it initially to divert and deliver 30 MG per day to the City and Del Este, and to provide for future expansion to 60 MG per day, upon construction of the necessary diversion and treatment facilities.

Both Del Este and the City have in fact experienced a decline in the groundwater level in the Modesto area, and degradation in its quality. Del Este states that its existing underground supply "has become inadequate for [Del Este's] needs without reference to any allowance for future growth."

(Application, p. 3.) Del Este has brought this situation to the Commission's attention several times since preparation of the

Montgomery Study, most recently in A.91-05-024. It now confronts what it calls a "critical" situation with respect to its underground supply. (Application p. 6.) An overdraft of the underground supply has existed for some time, and has been exacerbated by the ongoing six-year drought. "Put simply," Del Este states, "a treated surface supply is needed and needed now." (Application p. 6.)

The state of the s

A water supply study dated March 24, 1988, which was prepared for Del Este by Brown and Caldwell Consulting Engineers, states that tapping the surface supply available from the District under the terms of the MOU is economically the most favorable long-term alternative for Del Este's customers. It is also a matter of virtual necessity to do so, for Del Este's customers must be assured of a supply which will meet current California Department of Health (DHS) standards. As Del Este admits, "The present underground supply does not. The treated surface supply will." (Application, p. 7.) Del Este therefore believes that the Project must go forward, that Del Este's participation is essential, and that time is of the essence.

3. The MOU

The MOU provides that the District supply treated surface water to the City and Del Este by means of the Project, which consists of facilities necessary for the diversion, transmission, treatment, and delivery of 30 MG per day of surface water. The parties intend that all costs related to the Project be borne by

en en legende en legend

the state of the s

² See Re Del Este Water Company, D.90-12-053 (mimeo.), Re Del Este Water Company D.89-11-063, 33 CPUC 2d 508, and Re Del Este Water Company D.91-12-073 in A.91-03-024 (mimeo.).

³ Excerpts from three additional reports confirming the existence of the groundwater supply and quality problems and the need for the treated surface water project contemplated by the MOU are collectively included in the Application as Exhibit B.

the City and Del Este, with neither profit nor loss accruing to the District as a consequence.

8.81-18-20 - 0.85 - 0.87 Million (C. 18.48)

As described by the parties, the Project consists of a Design Phase, a Construction Phase, and a Commercial Operation Phase. The first phase is essentially complete, and the second may commence upon execution of the MOU. The purpose of the MOU is to memorialize the agreements among the parties with respect to the Project (including their respective obligations under the Design Phase), to provide funding by the parties for work done during the first phase, and to establish a framework for future agreements leading to the development, construction, and operation of the Project.

In the Design Phase, the District retained project engineers, environmental consultants, a financial advisor, an underwriter, and bond counsel. It also undertook to obtain permits, licenses, rights, and privileges necessary for construction, operation, and maintenance of the Project. The project engineers had to prepare preliminary design development plans and specifications, and a preliminary cost estimate for review and approval by the parties. After approval, the engineers will prepare final and complete construction documents and a final construction cost estimate. In the Construction Phase, financing and construction of the Project will be completed. Before commencement of this phase all necessary agreements, including a separate Treatment and Delivery Agreement (TDA), will be executed. Finally, the Commercial Operation Phase will begin upon commercial operation of the Project as described in the TDA.

The District will operate and maintain the Project and deliver water to the City and Del Este as provided by the TDA. Water delivered must meet all state and federal drinking water quality standards applicable to the Project. The first phase's costs and expenses will be included as an element of Project financing and will be reimbursed to District 70% by City and 30% by

applicant unless the City and Del Este agree to different proportionate shares. The MOU provides for reimbursement of a party's share upon abandonment or termination of the Project by one of the parties.

Del Este's future share of the first phase costs and expenses incurred by the District will be repaid to the District through the Project permanent financing. Del Este has already paid \$128,400 as its share (15%) of costs so far incurred by the District. The balance of the first phase costs (currently estimated to be \$5 million) will be incurred by the District in the future. Del Este believes that all of these costs will be repaid to District through permanent financing, and that its additional financial exposure upon execution of the MOU should not exceed \$1.5 million.

The MOU provides for approval of preliminary and final design work and cost estimates by the City, and Del Este, and requires the written authorization of both in order for the District to proceed under the construction contracts. The true interest cost of the debt financing is also subject to the approval of the City and Del Este. If the final financing terms exceed such true interest cost, the financing will not go forward.

The District and its staff are responsible for administering the MOU and carrying out the tasks necessary for the successful completion of each of the phases of the Project. While the District Board of Directors has the final decision-making authority with respect to the Project, the City and Del Este have the authority to approve or disapprove Project milestones, namely, engineering services, preliminary design and cost estimates, final design and cost estimates, and construction contracts. If either the City or Del Este fails to approve any of those milestones, it may terminate its involvement in the MOU and the Project. In the event of such termination, the terminating party must pay the

District its full share of Project costs to the date of

The MOU establishes two advisory committees to assist the District with the Project, a Policy Committee consisting of representatives of the District, the City and Del Este, which exercises review functions with respect to Project milestones, budget, and major design changes, and a Technical Committee consisting of one staff person appointed by each party, which has certain consulting functions with the Project manager relating to Project design, construction, operation and maintenance, and other advisory duties.

医乳腺 化对应 医透光 人名意格莱德格莱德克克克

The City and Del Este, through the Policy and Technical Committees, will participate in the annual budget preparation and approval process. Both committees have specified review and recommendation functions. If there is a dispute over the budget, the Policy Committee may submit an alternative to the objectionable budget item (or items) to the District Board, which will consider the alternatives and may adopt the annual budget only after public hearing. Either the City or Del Este may terminate the MOU prior to issuance of long-term financing if either determines that the Project is not feasible by reason of technical, engineering or economic reasons, that adequate insurance is not available at a commercially reasonable price, or that the District is not meeting its financing obligations or is failing to pursue those obligations with due diligence.

The parties expect to finalize the TDA after signing the MOU. The TDA will contain a "take or pay" undertaking, requiring the City and Del Este to pay stipulated percentages (70% and 30% respectively) of the fixed costs and debt service of the Project. This payment obligation will apply irrespective of whether the Project is fully operating or operable, but is subject to certain exclusions in the event that water is not delivered for 18 consecutive months for specified reasons other than drought. Delivered

Este's payment obligation to the District will include the District's actual cost of raw water at the same rate charged by the District to agricultural users, and the cost to the District of operating and maintaining the Project.⁴

SERVICES OF THE SERVICE OF THE SERVI

Del Este cannot state what its payment obligation under the TDA will be, since it will depend, among other things, upon the cost of the Project, the aggregate amount, term, and effective interest rate of the debt financing, and the costs of bond insurance, raw water, operation, and maintenance of the Project. However, Del Este believes that a reasonably accurate estimate of all those costs will be available before it enters into the TDA. Del Este states that it will not execute that Agreement unless it is satisfied that the projected costs are fair and reasonable, and until it has obtained appropriate Commission authorization.

4. Discussion and Conclusion and a second of the second of

Del Este's need for treated surface water to augment or supplant its groundwater supply is unquestionable. If Del Este fails to obtain an alternative source of supply, the depletion and degradation of groundwater, its only current supply, will continue in the face of Modesto's rapid growth. Moreover, there appears to be complete agreement that diversion of the District's surface water supply from agricultural to domestic use—at a time when agricultural land is rapidly being turned over to commercial and residential development—is the best available solution to Del Este's supply problem.

Del Este estimates that the cost of treated water from the Project will require an increase in its rates of approximately.

⁴ In addition to the foregoing, as security for its commitments under the TDA, Del Este will be required to pay the cost to the District (estimated at approximately \$40,000 per year) of bond insurance guaranteeing Del Este's payment obligation to the District.

200%. The Commission does not take lightly a request to approve an agreement that will have such a drastic impact upon Del Este's customers. But the request is justified: the higher cost of water is a price Modesto's ratepayers must pay for the region's explosive growth, which presumably brings with it corresponding ratepayer benefits.

If Modesto's needs could be met with a stable groundwater supply tapped by a limited number of wells, substantial investment in new diversion and treatment facilities would not be required. But the area has a finite supply of groundwater to meet its growing needs, and much of that is unusable because of the presence of contaminants and the strengthening of water quality standards. Since Del Este's share of the project cost will be paid for by rates charged after delivery of water from the new treatment plant, the cost of that treated water will be borne by the customers who use it.

The MOU contains adequate safeguards to insure that project costs will not be incurred imprudently, and that Del Este has a reasonable ability to control those expenditures. In this regard Del Este's rights coincide with those of the City, insuring consistency in the treatment of each by the District. Under these circumstances we authorize Del Este to enter into the MOU and to perform its obligations thereunder, and find that Del Este's execution of this agreement is reasonable and prudent and in the best interests of its customers.

As the approval herein affords complete relief to Del Este, and as the TDA is a separate document with entirely severable terms and conditions, we deny Branch's request to issue an interim decision. Consequently, this is a final decision. At such time as Del Este, the City, and the District finalize the terms of the TDA, Del Este may obtain authority from the Commission to execute that document by filing an advice letter. Branch will evaluate the

conditions of the TDA and provide its recommendation to the constant Commission in the form of a resolution.

- 1. Most of Del Este's water system, and about 80% of its customers, lie within the boundaries of the District.
- 2. Del Este's entire water supply for the portion of its system that lies within District's boundaries is currently drawn from underground.
 - 3. Del Este's groundwater supply is limited. The transfer and
 - 4. Del Este's groundwater supplies are declining.
- 5. The continued decline in the quantity of Del Este's product groundwater supply is resulting in a gradual degradation of the quality.
- 6. The DHS water quality standards have been revised since the 1984 Montgomery Study, are stricter than they were at the time that study was made, and are making it more difficult for Del Este to meet minimum water quality standards, particularly in relation to the dibromochloropropane content of its water.
- 7. The water supply and water quality problems which Del ...
 Este has been experiencing have been exacerbated by the ongoing drought conditions in its service area.
- 8. The District has a sufficient supply of surface water to divert 30 MG per day initially, and 60 MG ultimately, to Del Este and the City for municipal use.
- 9. The surface water which is available from the District for Del Este's and the City's use requires diversion and treatment before it can be delivered for such use, and facilities for this is purpose do not now exist.
- 10. Diversion and treatment of the surface water available from the District as provided in the MOU is the most economically favorable long-term solution to Del Este's water supply and water quality problems.

- 11. Treated surface water available from the District would meet all current DHS water quality standards.
- 12. The Stipulation for Settlement filed in this proceeding is reasonable in light of the whole record, consistent with law, and in the public interest.
- 13. As provided under the MOU, the Project consists of facilities for the diversion, transmission, treatment, and delivery of at least 30 MG per day of District's surface water supply to Del Este and the City.
- 14. The terms and conditions of the MOU are adequate to insure that Project costs will not be incurred imprudently, and that Del Este will have a reasonable ability to control those costs.
- 15. Del Este's execution of the MOU would be reasonable and prudent and in the best interests of its customers.
- 16. Since the MOU stands on its own and is severable from the terms of any related document to be executed by the parties in relation to the project, it is appropriate to issue a final decision in this proceeding at this time.

Conclusions of Law

- 1. Del Este should be authorized to execute the MOU, and to perform its obligations thereunder.
 - 2. The Stipulation for Settlement should be approved.
- 3. This is a final decision, and the proceeding is closed.

 This decision should be effective immediately, in recognition that the MOU has been executed by all of the parties thereto except Del Este, that there is an urgent need for the water to be supplied thereunder, and that time is of the essence in constructing the Project to obtain that water.

ORDER WAS ASSESSED FOR STATE OF THE STATE OF

The period was fined to now apply accorded 100 to the

- 1. Del Este Water Company (Del Este) is authorized to execute the Memorandum of Understanding (MOU) in the form attached as Appendix A, and to perform its obligations in accordance with the terms and conditions thereof.
- 2. Del Este may obtain authority from the Commission to execute the Treatment and Delivery Agreement (TDA) by filing an advice letter with the Commission at such time as Del Este, the City of Modesto, and the Modesto Irrigation District finalize the term of that agreement. Branch will evaluate the conditions of the TDA and provide its recommendation to the Commission in the form of a resolution. Samuel Broken and the Color of A
 - 3. The Stipulation for Settlement is approved. This order is effective today. Dated December 18, 1991, at San Francisco, California.

PATRICIAWM. ECKERT AND DATE. JOHN B. OHANIAN President -TO ANALYTY DANIEL WMT FESSLER GOODS OF NORMAN D. SHUMWAY Commissioners B. Will a late a factor of the control of the contr

မည္သည့္ ေရးကရည္သည္။ ၁၈၈ ျပည္သည္။ ရည္မ ေသးေတာင္းကားသို႔ မွာျမန္ The control of the co

ing the second bigother applied that along

- 5. 359/2015 tax = 2004/204 (4.7.4A)

CERTIFY THAT THIS DECISION WAS APPROVED BY THE ABOVE COMMISSIONERS TODAY ing for the market

MAN. Exocultive Director

ا در مورد و مداد داند. در معوره سوره از پههای داند. در موهد میان در است. در معواهد می داند محارز معی در از درای کود در کرد اداری میداند هگی در میشاند. در در در کا میان در کرد کرد می محافظ میک در در در در در در AMONG TELEPECTE ROLLINGT DIEECOM

ر ارتوان پر چرپههم هم خدون چې امال کاکياک د مولون هم اکياک د اماله امال کاکياک د اماله اماله اماله اماله ా - - - - - - - - గ్రామంలోని ధిరుందిరి ప్రాయేగరు మంచుకున్నాయి.మ 2.3. Problems of Bolimose in Estrement r - 1811 to demand the second was a size to a 2.7. Due of Avellande Warer to Carry of the Sycometal offer MEMORANDUM OF UNDERSTANDING AMONG

MODESTO IRRIGATION DISTRICT, CITY OF MODESTO

g in granuard installed to be The second secon A. II. GOVERN ON BEGINS The state of the s or a compared to the compared the compared to A.18. Project Manager A.21. Study గామ్మి. అయితులో డెలుకు 🕠 🕟 🕞 🤛 - wordingst builtakbor .co.A Androga Character and and because in the contract روب وهم معود العامل بينم يك يوي كمارات الكي∀د بيان بدايا معودات بدايات. اهل عراقي الأرام العاملين بين بي استان الكياب المائية المعاملات المعاملات المائية العامل العامل العامل العاملات

TO STAND SEED ON SERVICE OF STANDARD SEED ON STANDARD SEED ON 57980-16400/133/02/14/91/3

TABLE OF CONTENTS

MEMORANDUM OF UNDERSTANDING AMONG

MODESTO IRRIGATION DISTRICT, CITY OF MODESTO AND DEL ESTE WATER COMPANY

1.	PARTIES	1
2.		L
4 -	2.1. District Formation and General Purposes	L
	2.1. District formation and deneral response	1 2
	2.2. Water Management Study	2
	2.3. Problems of Reflance on Groundwater	•
	2.4. Water Rights	» 5.
	2.4. Water Rights 2.5. Study Recommendation 2.6. Drior Agreement of March 8, 1988	<u>ن</u> م.
		5
	2 7 Tich of Available Water To Carry Out Recommendations	
	of Study	3 3 3
	2.8. Enterprise Fund	3
	2.9. Purpose of This Memorandum	}
	•	
3.	AGREEMENT	ŀ
	DEFINITIONS	4
4.		4
	4.1. Advisory Committees	,, A
	4.2. Chief Executive Officers	1
		k ~
	4.6. Drainage Pumping	ł
	4.6. Drainage Pumping 4.7. Drainage Review Panel	•
	4.8. EIR	444555555555666
	4.9. Finance Committee	5
	4.10. Fixed Financing	5.
	4.11. Governing Bodies	5.
	4.11. Governing Bodies	5.
	4.12. Memorandum	5.
	4.13. Montgomery	Ś _
	4.14. Next Turnout	5
	4.15. Parties	ر د
	4.16. Policy Committee	-
	4.17. Project	<u>٠</u>
	4.18. Project Manager	Ò
	4.19. Service Area	5-
	4.20. Siting Study	5
	4.21. Study	6
	4.21. Study Costs	6
	4.23. Technical Committee	6
	4.23. Inclinical committees	6
	4.24. TIC	7
	4.25. Treated Water	7
	4.26. Treatment and Delivery Agreement	7
	4.27. Variable Financing	4
	4.21. Study 4.22. Sunk Costs 4.23. Technical Committee 4.24. TIC 4.25. Treated Water 4.26. Treatment and Delivery Agreement 4.27. Variable Financing 4.28. Water Review Panel THE PROJECT 5.1. Obligations of District and Expansion	
		- T
5.	THE PROJECT	_
	5 1 Obligations of District and Expansion	1

1.	5.2. Project Phases 5.2.1. Design Phase 5.2.2. Construction Phase	7
	5 2 1 Design Phase	7888888888
	5 2 2 Construction Phase	8
	5.2.3. Commercial Operation Phase 5.3. Area to be Served by Project	8
	r a bear to be Served by Project	8.
	5.4. District Sole Owner	8
	5.4. District Sole Owner 5.5. Disposition of Plans in the Event of Termination .	8
	5.6. Cooperation in Proceeding with Project	8
_	DESIGN PHASE 6.1. Design Phase Work 6.1.1. Consultants	9
6	DESIGN PHASE	9
	6.1. Design Phase Work	9
	DESIGN PHASE 6.1. Design Phase Work 6.1.1. Consultants 6.1.2. Property Interests	9
	6.1.2. Property Interests	10
•	6.1.3. Preliminary and Tinar Session 6.1.4. Environmental Clearance	10
	6.1.5. Preliminary Financing Work	10
	6 1 6 Negotiation of Treatment and Delivery	
	3 mm a m A M +	10
	co come dem phase I to be Advanced	ΤŌ
	6.2. Costs for Phase I to De Chanced Phase I Costs	10
	6.3. Reimbulsement of Advanced Line	_
_	CONSTRUCTION PHASE II	10
7 -	CONDUITOUR TITLE	
_	COMMERCIAL OPERATION PHASE III	11
8.	COMMERCIAL OPERATION PARSO	. 11
	8.1. Commencement of Phase III 8.2. District Obligations During Phase III	11
	8.2. District Obligations During Phase 111	11
	8.3. Standards of Treated Water	
	of a philapping of City and DEW to Take water	
•	o c rypansion of Groundwater Capacity and commingering	
	and the contract of the contra	
	8 6 1 Delivery	. 12
	O C D Cocke	. 13
	8.0.2. COSCS	. 13
		13
	8.6. Exchange of Groundwater for Treated Water 8.6.1. Delivery 8.6.2. Costs 8.6.3. Records 8.7. Force Majeure	
	ADVANCE OF PRE-CONSTRUCTION COST	. 15
9.	ADVANCE OF PRE-CONSTRUCTION COSTS.	
	9.1. Advances by District, City, and DEW, Sunk Costs,	. 15
• .	and Approval of Other Costs	
	9.2. Reimbursement of Sunk Costs and Advances,	. 15
,	Percentages	. 16
	9.3. Interest on Advances	
•	A A A AS CHARLOS AND NOCOTISTION MOTHE DV DEVH	
•	Party	. 17
	and the control of th	
10.	PROJECT FINANCING	. 17
	in i bistrict to Arrange Financing, Take or Pay	
	Provision Excuse from Take OF Pay	. 17
	10.1.1. Excuse, Fifty Percent of Contracted Water	
	not Delivered	. 18
÷	10.1.2. Excuse, Eighty Percent of Water Treatment	
	IU.I.Z. EXCUSE, Elgitty reflected of matter areasiment	. 18
	Plant Destroyed	_
	10.1.3. Excuse, Fifty Percent of Water Supply	
	10.1.4. New Period of Excuse	. 19
	10.2. Variable Costs	. 19
	10 3 City's and DEW's Payment Obligations, Treatment	

	3	
	and Delivery Agreement	19
•	and Delivery Agreement 10.3.1. Raw Water 10.3.2. Percentages 10.3.3. Records 10.4. Security for Payment, Rate Covenant by City 10.5. Drainage Pumping 10.5.1. Cost Items	19
	10.3.2. Percentages	19
	10.3.3. Records	19
, ,	10.4. Security for Payment, Rate Covenant by City	20
	10.5. Drainage Pumping	20
	10.5.1. Cost Items	21
-	10.5.2. Inability to Now Make Determination	21
	10.5.3. First Five Years	21
	10 5 A Tayogation of Paviou	21
	10.5.4. Invocation of Review	2.1
	10.5.6. Drainage Review Panel	21
	10.5.6. Diainage Review Panel	2.2
	10.5.7. Determination by Drainage Review Panel	22
	10.5.8. Findings and Decision	22
•	10.5.9. Subsequent Determinations	24
	10.6. Drainage Pumping in City and DEW Service Areas	24
	10.7. Marketing of Water Resulting from Drainage Pumping	24
	10.7.1. Allocation	24
	10.7.1. Allocation	24
-	10.7.3. Water Review Panel	40
	10.7.4. Determination by water Review Panel	40
	10.7.5. Findings and Decision of Water Review Panel	26
		•
11.	DELIVERY OF WATER 11.1. Water Supply to City and DEW 11.2. Formula for Water Entitlement 11.3. Level Deliveries 11.4. Drought, Proportional Curtailment 11.4.1. Curtailment 11.4.2. Conjunctive Use 11.5. Drought 11.5.1. Equal Priority 11.5.2. Adjustment of Curtailment	26
	11.1. Water Supply to City and DEW	26
1	11.2. Formula for Water Entitlement	26
	11.3. Tevel Deliveries	27
	11 4 Drought Proportional Curtailment	28
	11.4.1 Diought, Floportional cartaliment	28
	II. 4. I. CUITCALIMONG	28
	11.4.2. Conjunctive Use	70
	11.5. Drought	28
	11.5.1. Equal Priority	28
	11.5.2. Adjustment of Curtailment	28
	TT:	
	11.6. Water to Remain in Irrigation District Boundary .	29
12.	PROJECT MANAGEMENT	30
	12.1. District's Board of Directors	
	12 1 1 Pinal Decicione	30
,	10 1 0 Design Manager	30
	12.1.1. Final Decisions	30
	12.2. Dutles of Project Manager	30
ř	12.2.1. Implementation of Memorandum	30
	12.2.2. Notice to Advisory Committees	30
	12.2.3. Specific Duties	31
	12.2.3.A. Work Plans	31
4	12.2.3.B. Progress Reports	31
	12.2.3.C. Administering Contracts	31
	12.2.3.D. Public Information	31
	12.2.3.E. Recommending Consultants	31
	12.2.3.F. Financial Report	31
	12.2.3.G. Other Duties	31
	12.3. City and DEW Approval of Project Milestones	31
	12.3.1. Engineering Services	32
	12.3.2. Review of Preliminary Design	32
	12.3.3. Acceptance of Final Decient	32 32 32
	12.3.3. Acceptance of Final Design	32
		بكان

12.4. Processing Milestone Approvals	32
12.4. processing miles conempted and processing the first section of the	-33
12.5. Project Financing Approvals	33
12.5.1. Fixed Financing	34
12.5.2. Variable Financing 12.5.2.A. Initial Financing 12.5.2.A. Initial Financing	35
12.5.2.B. Conversion During Construction	
17 5.7 H. CONVERSION DULLING COMMENCE TO THE	35
Period	
12.5.2.C. Conversion Upon Commercial Operation	36
12.6. Advisory Committees	36
12.6.1. Policy Committees 12.6.1. Policy Committee 12.6.1.A. Project Milestones	36
12.6.1.A. Project Milestones	36
12.6.1.B. Budget Functions 12.6.1.C. Major Design Changes	36
12.6.1.C. Major Design Changes	
in c i'm other Maior Decisions (40-000000 00	37
መስጠር በብር ነው። መስመር መመር መመር መመር መመር መመር መመር መመር መመር መመ	
	•
in c n h Conemitation Will Profect Menages .	37
to commendations to Policy Commendations	37
in a communica Relative to Minor Decisions	
/<5 000 00 to \$100,000.00)	38
	38
12.7. Budget Principles 12.7.2. Budget Process 12.7.2.A. First Budget	38 38
17.7.1. Dudde kammanan	′38
12.7.2. Budget Flocess	38
12.7.2.B. Technical Committee Review and	
	38
	33
12.7.2.D. Policy Committee Review	39
12.7.2.E. Consideration of Budget by Policy	
12.7.2.E. Consideration of Budget by Fortage	40
Committee	
12.7.2.F. Rovision of Budget, Alternate Budget	40
and Report	41
12.7.3. Public Hearing	4.1
$oldsymbol{a}$	41 41
12.7.4. Appearance at neutring 1	~ 1
13. TREATMENT AND DELIVERY AGREEMENT	4.1
12 1 Negotiation	42
13.4. Further Detail in Treatment and Delivery Agreement	4.2
المراجع والمحاربة والمناز والم	
14. LIABILITY 14.1. Insurance, General	42
14.1. Insurance, General	42
14 1 1 COMMETCIAL GENERAL PLADITICA OF COMPAC	
hansive General Liability Insurance of the control	43
14 1 2 Comprehensive Automobile Liability	
Insurance	43
14 1 3 Umbrella or Excess Liability Insurance	43
14 1 4 Workers Compensation Insurance	43
14 1 5 Property Insurance	77
14 1 6 Insurance Provisions	77
14 1 6 % Cancellation	44
14 1 K.B. District and others to be Named	44
14.1.7. Approval of Insurers	77
14.1.7. Approval of institute 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	44

Commence of the second

	• • •	14.2. Indemnification and Hold Harmless of the page 4	5
•		14.3. Third Party Claims, Not Covered by Insurance,	
	•	Willful	2
		14.4. Third Party Claims, Not Covered by Insurance,	1
		Ordinary (Sec.) and a subject to the control of th	
		14.5 Claims Between Parties 4	
		14 6 Workers' Compensation Claims 4	7
		14.7. Replacement of Pro Rata Right of Contribution 4	7
	15.	RELATIONSHIP OF PARTIES	7
	16.	GENERAL PROVISIONS GOVERNING MEMORANDUM	3
	17.	GENERAL PROVISIONS GOVERNING MEMORANDUM 49	9
		17.1. Severance	9
		17.2. Waiver	0
		17.3. Counterparts 50)
		17.4. Supporting Resolutions and PUC Approval 5	5
		17.5. No Rights in Other Parties	5
		17.6. Amendment A.	1
		17.7. Obligations Prior to Termination	1
		17:7: Obligations Plant to Telmandelough, a traction of	ī
		17.8. Captions	ī
		17.10. Each Term Independent	•
		17.11: Governing Law 2.2222	•
		17.11. Governing baw and an analysis and a second s	-
	10	TERM AND TERMINATION	1
	18-	18:1. Execution by All Parties C. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	- 1 4
		18.1.1. Superseded by Other Agreement	
		18.1.2. Termination by Unanimous Agreement 5	2
		18.2. Termination of Participation Prior to Financing . 5	_
		- 1	2
	19.	UNDERTAKINGS 5:	,
		WATER RIGHTS AND OWNERSHIP	•
	20.	WATER RIGHTS AND OWNERSHIP 5:	2
	21.	NOTICES	3
	· · · · · · ·	EXHIBITS.	
	1 .		
	Exhi:		
•		RESOLUTIONS AND APPROVALS	
		lution of City Authorizing Execution	
		lution of DEW Authorizing Execution	
Ċ	Cali	fornia Public Utilities Commission Approval	
-			
	· · · · · · · · · · · · · · · · · · ·		
,		ကြောင်းသည်။ မေးကြောင်းသည်။ ကောင်းသည် မေးကြောင်းသည် မေးကြောင်းသည်။ မေးကြောင်းသည် မေးကြောင်းသည်။ မေးကြောင်းသည်။ မေးကြောင်းသည်။ မေးကြောင်းသည်။ မေးကြောင်းသည် မေးကြောင်းသည် မေးကြောင်းသည်။ မေးကြောင်းသည်။ မေးကြောင်းသည်။ မေးကြော	
•	• • •		d
		The state of the s	
•	•	the control of the co	•
•		မ်ားမှ သန်လလည်း သည်သည်။ ကြောင်းလည်းမှာတွေကြာကြာလည်းများများများများများများများများများမျာ	

The second of the control of the con

and the second of the second o

en alle transport de la company de la compan

کا بمیں داخت بیٹر پہلائا کا عمر رست ابو کو ہو جو بعد اور کا در ایک کا بہت ہا کہ ایک اور کی است. پر پردر اور کریں جو کا جو در افراد کی اور اور ایک اور در ایک در ایک در ایک در ایک در ایک در است کا میں در ایک

Corrected Copy 02/14/91 Revision To the ledbalm on this weigted odd to

THE LANGE CONTROL OF THE CONTROL OF THE PROPERTY OF THE PROPER

AMONG

MODESTO IRRIGATION DISTRICT, CITY OF MODESTO

AND DEL ESTE WATER COMPANY CONTROL OF THE CONTROL OF T

1. PARTIES.

The Parties to this Memorandum of Understanding ("Memorandum") are the Modesto Irrigation District, a California irrigation district ("District"), the City of Modesto, a California municipal corporation ("City"), and Del Este Water Company, a California corporation ("DEW").

2. RECITALS.

and the second of the second of the second

This Memorandum is made with reference to the following that and circumstances, among others:

2.1. District Formation and General Purposes. District was ు కుండా కాడుకు ఉందుకున్న ఉందార్థుడి ఉంది. అడి formed more than 100 years ago for the purpose of providing า เรื่องสาพารเรียกหญางเรื่อง เรื่องที่สมันแดน กล้าสานตากหลัง irrigation water to serve the needs of agriculture. To that end, معملون کی مقاولان کے دواروں کے اور دیا کہ ان کے ان میں ایک کے ان اور ان اور ان اور ان اور ان ان اور ان ان اور میں موجود کی اور داخل کی ان میں کا ان ان کی ان میں کو ان ان کی ان ان ان کی ان ان ان کی ان ان کی ان ان کی ان کی District has perfected water rights sufficient to meet those ្នាស់ សំខេត្ត ខេត្ត ខ្លែង សំណើង និងជា ស្រាយស្រាស់ ប្រាស្មានស្ថិតនៃក្រុម ប្រាស់ និងក្រុម ប្រែក្រុម needs. District's development of its water rights and its agricultural irrigation delivery system, and the careful atten-الإستان الموافق الموافقة في الموافق المتابعة والمتابعة والمتابعة المتابعة المتابعة المتابعة المتابعة المتابعة الإستان المتابعة في المتابعة المتابعة المتابعة المتابعة والمتابعة المتابعة المتابعة المتابعة المتابعة المتابعة tion to the needs of District's irrigators, has been prominent in medical design of the control of the making Stanislaus County one of the world's foremost agricultural Company of the property of the first with areas. In recent years, population growth within District's వ్రామాత్రంలో కాట్లుకోతే 👍 🗘 🗘 🕠 ا ہے۔ فاگ مختلف سے آپ انسون پرمسے در در ایک ایک انسون کا کوف بھائے انتظام کی انتظام کا کا انتظام کا boundaries has reduced the irrigated acreage while increasing the demand for municipal uses of water. The Parties desire to put to beneficial municipal use for the inhabitants of District certain

amounts of surface water to which District has rights. In doing so, the Parties shall be mindful of the great importance of District's water rights, and the significant role of agriculture within District. The portions of the Service Areas of City and DEW to be served with water pursuant to this Memorandum are within the Irrigation District Boundary of District. The water to be supplied by District pursuant to this Memorandum is not surplus water, but rather is water which has historically been put to beneficial use within District and is necessary for use within District.

- 2.2. Water Management Study. District and City authorized, received, and accepted a Water Management Study prepared by James M. Montgomery Consulting Engineers, Inc., dated November, 1984.
- 2.3. Problems of Reliance on Groundwater. The Study found, in part, that existing ground water sources of City and DEW are limited and supplies are declining, and that a continued decline in the quantity of ground water will result in a gradual degradation of the quality of the ground water. This degradation of quality, combined with on-going modifications to State and Federal drinking water standards, will make it increasingly difficult to meet domestic water supply demands solely through the use of ground water.
- 2.4. Water Rights. District has certain water rights as to waters of the Tuolumne River.
- 2.5. Study Recommendation. The Study recommends, in part,

 that City and DEW contract with District for treated surface

 water to supplement their ground water supplies.

., *, ,...

- 2.6. Prior Agreement of March 8, 1000. District, City, and DEW entered into an agreement dated March 8, 1988 to continue feasibility studies and preliminary design of the Project which includes siting of the Project through work by Montgomery, authorized by District Resolution 86-146, and by URS Consultants, Inc., authorized by District Resolution 88-84 and as amended by District Resolution 89-36.
- Study. It is in the best interest of the Parties and their respective water users to utilize District's available water to give effect to the recommendations of the Study in a manner which is timely and reasonable, which is environmentally acceptable, and which is not injurious to agriculture or other interests of District.
 - 2.8. Enterprise Fund. It is the intention of the Parties that, except as specifically provided otherwise herein, the Project shall be financially operated as an enterprise fund with costs borne by City and DEW and with neither profits nor losses accruing to District as a result of the Project.
 - enter into this Memorandum of Understanding in order to memorialize their agreements with respect to the Project including
 obligations under the Design Phase (Phase I) of the Project as
 set forth in Section 6; to provide funding by the Parties for
 work done during Phase I; and to establish a framework for future
 agreements leading to the development, construction, and operation of the Project and the rights, responsibilities, duties, and

obligations of the Parties with regard theretower and

- 34 FOR AGREEMENT. Control of the control of the control of the WET
- The warm Theo Parties agree was of ollows: xq box gold box gold idd a sold
 - 4.00 DEFINITIONS PRODUCT OR SEASON SEASON OF THE PARTY AND ASSESSED AS
 - to Section 12.6.
 - 4.2. Chief Executive Officers: The City Manager, the President of DEW, and the Chief Executive Officer of District, or their respective designees.
 - 1 0 04.31 City: City: of Modesto. programmer percentage end processes
 - 1 4.4. DEW: Del Este Water Company of the state of the st
 - 4.5. District: Modesto Irrigation District.
- 4.6. <u>Drainage Pumping</u>: Any pumping, including needed wells and pumps and other capital improvements or the use of existing facilities, undertaken by the District to remove water from the surface or subsurface areas in order to protect the viability of crop, including orchard crop, production and in order to prevent interference by the water with structures or utilities, if the pumping is necessitated as a result of rising water tables resulting from the Project other than within the Service Areas of City and DEW.
 - 4.7. Drainage Review Panel: The Panel which may be established from time to time pursuant to Section 10.5.6.
 - 4.8. EIR: The Final Environmental Impact Report for the Modesto Surface Water Treatment Plant, prepared by USR Consultants and dated March 1990, a copy of which is on file with District.

The second secon

The second of th

- Director of Finance of City, the Chief Financial Officer of District and a person designated by the Board of DEW who is experienced and knowledgeable as to the financial affairs of DEW, or their respective designees.
 - 4.10. Fixed Financing. Long Term Project financing with fixed terms and rates extending for the length of the repayment period.
 - 4.11. Governing Bodies: District's Board of Directors,
 City's City Council, and DEW's Board of Directors.
 - 4.12. Memorandum: This Memorandum of Understanding between District, City, and DEW.
 - 4.13. Montgomery: James M. Montgomery Consulting Engineers,
- 4.14. Next Turnout: The first point of diversion onto private land or into a smaller canal or pipeline below any point at which City or DEW discharges groundwater into an irrigation canal as provided in Section 8.6.
 - 4.15. Parties: District, City, and DEW.
 - 4.16. Policy Committee: The Committee created pursuant to Section 12.6.1 of this Memorandum.
- 4.17. Project: Alternative "A" as described in the EIR and in the adopted project described in Resolution No. 90-50 a copy of which is attached hereto as Exhibit "A". Project also includes land and right of way acquisition, environmental mitigation, including, but not limited to, Drainage Pumping established as necessary pursuant to Section 10.5, facilities, appurtenances,

State of the state

Section 1981

except the obligation of Paragraph 8.5, and any other actions, including the defense or prosecution of litigation, necessary to the completion and operation of the Project. The nature and description of the Project may be changed and refined over time by District, subject to the approval of City and DEW and subject to compliance with applicable provisions of this Memorandum and the Treatment and Delivery Agreement. The Project shall initially have a design capacity of 30 million gallons per day. At the time, and under the terms, as the Parties agree the Project may be expanded, provided, however, that the Project shall in no event exceed actual production of 60 million gallons during any day.

- 4.18. Project Manager: Such person or entity as may be appointed pursuant to Section 12.1.2 of this Memorandum.
- 4.19. Service Area: The actual area within the District to be served by the Project as described in Section 5.3; to which, at any given time, City or DEW, respectively, is serving water.
- 4.20. Siting Study: The Modesto Surface Water Treatment Study, prepared by Montgomery, dated March, 1989.
- 4.21. Study: The Water Management Study, prepared by Montgomery, dated November, 1984.
- 4.22. <u>Sunk Costs</u>: Those costs associated with this Project incurred by District prior to the execution of this Memorandum, as the costs are set forth in Exhibit "B".
- 4.23. Technical Committee: The committee created pursuant to Section 12.6.2 of this Memorandum.
 - 4.24. TIC: "True Interest Cost" as the term is commonly

الرواقة المروح وهوم وقوارة في يستويد الواد المستويد الرواد المستويد الأن المستحد في تراسيد الأن الأن الرواقة ا الأنظام والرواقة المستويد الرواقة الرواد المستويد المستويد المستحدة المستحدة الأن المستحد المستويد المستويد ال

- : --

applied in municipal finance.

- 4.25. Treated Water: All water delivered from the Project to City and DEW at the point or points of delivery.
- 4.26. Treatment and Delivery Agreement. The agreement or agreements among District, City, and DEW, or any of them, entered into pursuant to Section 13 of this Memorandum.
- 4.27. Variable Financing: Project Financing with variable rates and terms during the construction phase and until Fixed Financing is put in place.
- 4.28. Water Review Panel: The Panel which may be established, from time to time, pursuant to Section 10.7.3.

 5. THE PROJECT.
- design, finance, construct, operate and maintain the Project for the purpose of treatment and delivery of water to City and DEW. The location, configuration, design, construction, and financing of the Project will be determined in accordance with this Memorandum, and subsequent agreements among the Parties hereto, all as set forth herein. In the event the Parties agree upon an expansion of the Project beyond a design capacity of 30 million gallons per day, the expansion shall not take place until the Parties have executed all agreements necessary to undertake and complete the expansion, compliance with all applicable environmental laws and regulations, obtaining all required licenses and permits, and obtaining financing.
- 5.2. Project Phases. Completion of the Project shall be undertaken in three (3) phases:

And the second

ا من المحادث الله المراجعة الأحماد المراجعة والمحادث المحادث المحادث المحادث المحادث المحادث المحادث المحادث ا المراجعة المحادث المحادث

- 5.2.1. Design Phase: (Phase I); and
- 5.2.2. Construction Phase: (Phase II); and
- 5.2.3. Commercial Operation Phase: (Phase III).
- 5.3. Area to be Served by Project. The area to be served by the Project is depicted in Figure 2-1 of the EIR. City and DEW shall have the right to determine where within the area depicted in Figure 2-1 of the EIR, and where within their respective Service Areas, to use the water from the Project, provided, that, in no event shall water from the Project be used outside the Irrigation District Boundary of District (as opposed to the Electrical Service Area Boundary of District).
- 5.4. <u>District Sole Owner</u>. District shall be the sole owner of the Project.
- 5.5. Disposition of Plans in the Event of Termination. The strategical straining the event that any Party terminates or abandons the Project, DEW, ما المعادل في المعادل الموجد الموجد والموجد والمعادل المعادل المعادل المعادل المعادل المعادل المعادل المعادل ا المحادث المواجد المعادل المعاد and City, respectively, shall be entitled to obtain copies of all reports, drawings, studies, plans, specifications, other engimy with neering documents, and all other documents pertaining to the Project, provided that DEW or the City, as the case may be, has కూడా చేసాకానకి మేట్లు ఇంద fully paid its share of all costs due up to the time of the termination or abandonment as those costs may be due and owing The state of the s ార్ట్ స్ట్రైమ్ కార్డ్ ఉన్నాయినే నటి pursuant to this Memorandum or any Treatment and Delivery Agree-্তি প্ৰত্যাহ্য ক্ষিত্ৰত সাধাৰত্বত প্ৰত্যালয় সূত্ৰত্বত প্ৰত্যালয় স্থান ment. ్రాగ్ క్రిక్స్ క్రామంలో కథ్యక్షు కోర్యాయ్డ్ క్రామంలో కోట్లుకోన్నాయి. అయిని తెలికా ఫ్రోమాన్స్ క్రామం
 - 5.6. Cooperation in Proceeding with Project. District shall, to the extent it has not already done so, utilize its best efforts to acquire and maintain ownership of all easements, water conveyance rights, water rights, federal and state licenses, and

permits, and all other rights, entitlements, and privileges necessary for the construction, operation, and maintenance of the Project. City and DEW shall utilize their best efforts to promote the Project and its financing and assist the District in regard to the matters set forth in this Section 5.6 in order to facilitate the financing, construction, operation, and maintenance of the Project.

6. DESIGN PHASE.

6.1. Design Phase Work. To the extent items are not already completed or underway, Phase I work shall begin upon execution of this Memorandum by all Parties. During Phase I, each of the following shall occur:

interpretation in an engineering and in a con-

- engineer or engineers, an environmental consultant, a financial advisor, underwriter and bond counsel, and shall also to the extent practical during Phase I obtain all licenses, permits, entitlements and privileges necessary for the construction, operation and maintenance of the Project.
- land, easements, and rights of way required for the construction, operation, and maintenance of the Project other than real property interests already owned by any of the Parties. City and DEW shall assist with the identification and acquisition of all water storage sites within their respective Service Areas. The Parties agree that they will use their best efforts to place water storage facilities within existing property already owned by one of the Parties.

- engineer or engineers shall prepare preliminary design development plans and specifications and a preliminary cost estimate, and, after approval by the Parties of the preliminary design and cost estimate as provided in Sections 12.3 and 12.4, the engineer or engineers shall prepare final and complete construction documents and a final cost estimate.
- 6.1.4. Environmental Clearance. The environmental consultant and District shall take all action necessary to comply with all applicable environmental laws.
- 6.1.5. Preliminary Financing Work. The financial advisor, underwriter, bond counsel and District shall perform preliminary work necessary for the financing of the Project and shall make a financing proposal to City and DEW.
- 6.1.6. Negotiation of Treatment and Delivery Agreement.

 The Parties shall negotiate and approve a Treatment and Delivery

 Agreement in accordance with, and in furtherance of, this Memorandum.
- 6.2. Costs for Phase I to be Advanced. Costs for all work performed during Phase I shall be advanced in accordance with the provisions of Section 9.1.
- 6.3. Reimbursement of Advanced Phase I Costs. Reimbursement of the costs advanced pursuant to Section 9.1 shall be made in accordance with Section 9.2.
- 7. CONSTRUCTION PHASE II. To be seen to be year and source

Phase II shall consist of finalization of the financing and the construction of the Project. Prior to the commencement of

Phase II all necessary agreements for the financing, construction, purchase, and sale of water, and operation and maintenance of the Project including, but not limited to the Treatment and Delivery Agreement, will be executed.

8. COMMERCIAL OPERATION PHASE III.

- 8.1. Commencement of Phase III. Phase III shall commence upon completion of the construction of the Project, as completion is defined in the Treatment and Delivery Agreement described in Section 13, and final acquisition of all the permits, licenses, entitlements, rights, and privileges required to operate and maintain the Project.
- 8.2. <u>District Obligations During Phase III</u>. During Phase III, the District shall operate and maintain the Project, and shall deliver Treated Water to the City and DEW consistent with the terms of the Treatment and Delivery Agreement.
- 8.3. Standards of Treated Water. Throughout the term of the Treatment and Delivery Agreement, or any successor or substitute agreement, or extensions thereof, District shall deliver to City and DEW Treated Water which meets all state and federal drinking water quality standards applicable to the Project at the time of delivery to the City or DEW system.
- 8.4. Obligation of City and DEW to Take Water. City and DEW at all times shall exercise their best efforts to take all Treated Water made available to them.
- 8.5. Expansion of Groundwater Capacity and Commingling of Groundwater. Throughout the term of the Treatment and Delivery Agreement, City and DEW shall maintain and expand their respec-

tive well systems so as to be able to meet their own customer demand as they deem necessary. Groundwater will be delivered by City and DEW through their respective same water pipelines which will carry the treated surface water supply provided by District, resulting in a physical commingling of the groundwater and surface supplies.

- 8.6. Exchange of Groundwater for Treated Water. Subject to THE PROPERTY OF A STATE OF THE PROPERTY OF THE the absolute limitation of 60 million gallons per day set forth in Paragraph 4.17, and subject to approval of District as to time ్రాముల్లో కాటుక్రాము. మాట్లు అయిన స్టామ్ ఉంది. ఇంటు and place of delivery, which approval shall not be unreasonably withheld, City and/or DEW, at their respective options, may and and Table and Albania deliver groundwater to the District irrigation canal system in exchange for an additional amount of surface water from the District equal to the quantity of groundwater delivered to the ంచే కృత్యంత ప్రభామంత్ర జంత్రిమేంద్ర మహిందు District canal; provided that each of the following conditions are met: ing a single of the control of the c
- 8.6.1. Delivery. Any groundwater so delivered by City ్రామాని గవరాళు పరుళ్ళ కూడారు. జీవరాణుజీవరం or DEW shall be delivered into one of the irrigation canals of ระวัง เวลารายลัส เพละ และการหลา ส**อ**ามุสสอดราชก District. After blending the groundwater discharged with the ar his sprake that a sprott bat by water then in the canal, the quality of the water, sampled at the Next Turnout, shall be of a quality suitable for agricultural use, including without limitation, use for crops, orchards or livestock. In the event that water samples at the Next Turnout The said of the sa demonstrate, based upon water quality standards applicable, to The state of the s the use of the water for agricultural purposes, that the quality of the groundwater after mixing is not suitable for agriculture, City or DEW shall immediately cease making those discharges which Action of the composition of the

contribute to the unsuitability of the water at the point at which the sampling occurred.

- 8.6.2. Costs. The Party seeking exchange pays for all costs, including additional capital facilities, if necessary, associated with delivering the groundwater supply to the District irrigation canal system, subject to the same costs and delivered in the same manner as water otherwise delivered by District to City and DEW pursuant to this Memorandum but without a charge for raw water.
- 8.6.3. Records. The Party seeking exchange maintains a record of the quantity of groundwater delivered to the District irrigation canal system and the quality of blended water in the canal below each point of introduction of groundwater.
- 8.7. Force Majeure. District shall be excused from its obligation to deliver Treated Water in the event that District is rendered unable, wholly or in part, by force majeure to carry out its obligations under the Treatment and Delivery Agreement. Upon the occurrence of an event of force majeure, District shall give notice and full particulars of the force majeure in writing, or by telephone followed by a writing. District's performance shall be suspended during the continuance of the force majeure. The term "force majeure" as used herein shall mean acts of God, strikes, lockouts, failure or refusal of any person or entity to comply with then existing agreements to obtain or ship materials or equipment, or industrial disturbances, acts of a public enemy, wars, blockades, insurrections, riots, epidemics, landslides, lightning, earthquakes, volcanic eruptions, fires, flood, wash-

outs, or other natural disasters, threat of physical harmore damage resulting in the evacuation or shutdown of facilities necessary for the supply, treatment, and distribution of water, arrests and restraints of governments and people, civil disturbances, insurrection, explosions, sabotage, restraint by court order or public authority, other than the District, having jurisdiction over the Project, and action or nonaction by, or failure to obtain authorizations or approvals from, any governmental agency or authority of competent jurisdiction, and any other causes, whether of the kind herein enumerated or otherwise, not within the control of the Party claiming force majeure and which, by the exercise of due diligence, the Party is unable to prevent or overcome. District will use its best efforts to promptly bring to an end any condition falling within the definition of force majeure. It is understood and agreed that the settlement of strikes or lockouts shall be entirely at the discretion of the Party having the difficulty. In the event of any strike or lockout which may render District unable, wholly or in part, to carry out its obligations under the Treatment and Delivery Agreement, District shall promptly implement a plan to reasonably ensure the continued operation of the Project and continued delivery of Treated Water pursuant to the Treatment and Delivery Agreement. Furthermore, in the event of a strike or lockout, District's performance under the Treatment and Delivery Agreement will not be suspended for a period of more than 60 days. If the District gives notice of a force majeure event which impacts the District's water deliveries, then the provi-

sions of Section 10.1 concerning nonpayment by the Cityland DEW for failure to receive their full water entitlement shall still be applicable.

- 9.1. Advances by District, City, and DEW, Sunk Costs; and Approval of Other Costs. City and DEW shall be liable for the payment of all costs and expenses of all Phases of the Project in accordance with this Memorandum and any subsequent Treatment and Delivery Agreement. To facilitate the planning and construction of the Project, District has already advanced certain Project costs and unless otherwise agreed District, City, and DEW will advance all costs reasonably necessary for completion of all engineering and design work, feasibility studies, environmental review, permit, and licensing costs, and all other work conducted during Phase I of the Project in the following portions: District, 50%; City, 35%; DEW, 15%. Such costs shall be reimbursed by City and DEW as set forth in Section 9.2; however, in order for costs incurred by District to be reimbursed by City and DEW, the costs must be either included as Sunk Costs identified on Exhibit B or approved by City and DEW. Approval shall not be required as to costs incurred as a part of a Project milestone as provided in Sections 12.3 and 12.4. Any cost not approved by City and DEW shall be subject to further review in the manner provided in Section 12.6. As an argument where his roughnic
- 9.2. Reimbursement of Sunk Costs and Advances, Percentages.
 City and DEW shall each reimburse District for all of the Sunk
 Costs and all costs advanced by District pursuant to Section 9.1

based upon their respective proportionate shares of the Project. Unless a Treatment and Delivery Agreement has been executed and provides otherwise, or unless City and DEW have agreed and advised the District of different percentages which still total 100%, the respective proportionate share of City's and DEW's reimbursement obligation shall be City 70% and DEW 30%. Said reimbursement shall occur upon abandonment of the Project by a Party, termination of the Project pursuant to Section 18.1.2 or 18.2 of this Memorandum, or upon execution of the Treatment and Delivery Agreement, whichever first occurs, provided, however, that in the event the Parties execute a Treatment and Delivery Agreement as set forth in Section 13, City and DEW shall each be permitted, at the option of each Party, to provide for the repayment of the Sunk Costs and all costs advanced by District through the financing obtained pursuant to Section 10.1. In the event of a termination of the Project pursuant to Section 18.2, and all Parties abandon the Project, neither City nor DEW shall be required to reimburse District pursuant to this Section 9.2. If City and DEW, or either of them, proceed on the Project without District, and provided District has delivered to City and DEW all Project drawings, documents, permits and all other materials necessary to the completion of the Project, City and DEW shall reimburse the District for all costs advanced by District in their proportionate shares of the Project.

9.3. Interest on Advances. Amounts advanced by District pursuant to Section 9.1 shall bear interest at the same average monthly yield as District earns on its general fund portfolio,

provided, however, that if at anyotime the District shaverage cost of borrowing money exceeds the District share of areturn on its general fund portfolio, the interest contamounts advanced by the District shall be the District's cost of borrowed money. In no event shall the interest charged by the District pursuant to this Section exceed the legal limit. "Such sinterest shall taccrue monthly beginning at the time District actually expends the funds on which the interest is to accrue?

- 9.4. Costs of Studies and Negotiation Borne by Each Party.
 Costs of studies conducted by any Party for its own purposes,
 costs associated with the preparation and negotiation of this
 Memorandum or subsequent or other agreements among the Parties,
 including agreements relating to the delivery of water, shall be
 borne by the Party incurring the costs and shall not be advanced
 pursuant to Section 9.1, nor reimbursed pursuant to Section 9.2.
- Excuse from Take or Pay. The District, subject to Section 12.5, will use its best efforts to arrange financing for the construction of the Project using bonds or other evidences of indebtedness or certificates of participation, which will be secured in part by a Treatment and Delivery Agreement with City and DEW. The Treatment and Delivery Agreement will contain a "take or pay" provision which requires City and DEW to pay their respective agreed percentages of the fixed costs and debt service of the Project whether or not the Project or any part is operating or operable or its output or capability is suspended, interrupted,

The state of the s

e nambre agric a la liger anti-

interfered with, reduced or curtailed, or terminated in whole or in part. The payments shall not be subject to reduction whether by offset, counterclaim, recoupment, or otherwise and shall not be conditioned upon the performance or nonperformance by any Party to any agreement or for any other cause or reason whatsoever. The "take or pay" obligation of City and DEW with respect to debt service shall commence at the time that debt service payments actually commence under the applicable Project financing. The "take or pay" obligation of City and DEW with respect to fixed costs shall commence at the time of the commencement of Phase III. Neither City nor DEW shall be required to pay for water not delivered under the following specific conditions:

Delivered. At least fifty percent (50%) of the City's and DEW's contracted water is not delivered for a period longer than eighteen (18) consecutive months for reasons other than drought.

Destroyed. In the event that eighty percent (80%) or more of the water treatment plant which is a part of the Project is destroyed or disabled the eighteen (18) month period shall be extended to twenty-four (24) months.

event that fifty percent (50%) or more of the District's total annual water supply, as adjusted as provided in Section 11.2, is lost or unavailable for physical reasons other than drought, the eighteen (18) month period shall be extended to twenty-four (24) months.

- Tage, 25 m bar grant (las) compared to a Page, 25 m bar grant (las) compared 10.1.4. New Period of Excuse. A new eighteen (18) month, and, if applicable, twenty four (24) month period, shall not occur for the purpose of this Section 10.1 until a twelve CLASSON TO THE CONTROL WAS A STORY TO THE (12) month period, during which at least eighty percent (80%) of the City and DEW's water requirement has been delivered, has and the second of the second o passed. ్రామంగా కార్యాల్ గ్రామంలో హిందు క్రామ్మిక చెప్పుడు మీదుగ
- 10.2. Variable Costs. The City and DEW will also be bound to juli pa katroskupab abawas ericania. Antonio del Carlo de to pay for the variable costs of the Project. The portion of variable costs payable by City and DEW shall be in the same proportions as the quantity of water each Party receives during the accounting period.
- 10.3. City's and DEW's Payment Obligations, Treatment and to be not in the first of the section والأمراطية فالمتراوية فالأعراز الهالك Delivery Agreement. City's and DEW's payment obligations to District, will be refined and set forth in the Treatment and و مراجع بعد بعد المراجع من الرحم بعد المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع ا المراجع المراجع المراجع بعد محمد بعد مراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع الم Delivery Agreement, and shall include all of the following components: දිය වන අදුරුදුව කරන වන අතුරු කියල් කියල් විසින්ට ද මුත්තිකියම් කියල්
- 10.3.1. Raw Water. The actual cost of raw water which is treated and delivered, with the cost calculated at the same rate charged by the District to agricultural users. రాజా కున్నా హోక్షకాత్ర కారా గ్రామానికోడ్డా తంచా చూపేత్స్కుడు.
- 10.3.2. Percentages. City's and DEW's respective The second section of the second seco shares (70% and 30%, respectively, unless City and DEW agree to a different split of the 100% repayment obligation) of the Project financing payments as previously approved by City and DEW.
- 10.3.3. Records. City's and DEW's respective shares of the costs to District of operating and maintaining the Project, provided that District shall (i) keep and maintain and provide to ುಜ⊝ಕಚರ √ಫಾರಣಕ್ಷ್ City and DEW detailed cost accounting reports documenting the

ALCOHOLD CONTRACTOR

- costs, (ii) keep and maintain separate accounting and bookkeeping records with a separate account and fund for the Project, and (iii) allow City and/or DEW and their employees, accountants, attorneys and agents to review, inspect, copy and audit the accounting and bookkeeping records, including all source documents. District shall have the right to review, inspect, copy and audit all accounting and bookkeeping records, including all source documents of City and DEW as may pertain to the receipt, delivery and sale of water received from the Project.
- 10.4. Security for Payment, Rate Covenant by City. Prior to obtaining the financing as set forth in Section 10.1, District shall be entitled to the assurances it may reasonably deem necessary, and be entitled to the financial information as may be necessary, to ascertain that each Party is in the financial condition as will allow the Party to fulfill its financial commitments to the Project. In the event that District deems necessary, District may require security for the commitment, including, but not limited to, bond insurance, letters of credit, The second of the form of the second of quarantees, cash deposits, and/or bonds. Such security shall be The second secon subject to approval of existing mortgagors, lenders, bond holders, and/or the California Public Utilities Commission as necessary. It is agreed by the Parties that City's security obligation pursuant to this Section 10.4 shall be satisfied by a rate covenant in a form satisfactory to District.
- DEW shall include the cost of Drainage Pumping only if, and only to the extent, determined and provided in this Section 10.5. as

· follows: Down to the letter of the land of the particular and book

include capital, operations, maintenance, and replacement items.

Parties agree that at the time of entering into this Memorandum and the Treatment and Delivery Agreement they are not able and will not be able to determine whether Drainage Pumping will be necessary as a part of the Project nor even the proper basis on which to make a determination.

10.5.3. First Five Years. No Drainage Pumping costs shall be paid by City and DEW during the first five years of the Commercial Operation Phase or as a result of operations during the first five years of the Commercial Operation Phase.

of the fourth year of the Commercial Operation Phase, or during the first month of each successive year of the Commercial Operation Phase, District may initially invoke a review of whether Drainage Pumping is required as a result of the Project.

two months following the invocation of a review, the chief executive officers of the Parties shall meet and endeavor to agree upon the determinations to be made by the Drainage Review Panel pursuant to Section 10.5.7 and 10.5.8 of If the chief executive officers reach unanimous agreement, then their determinations shall apply as if made by the Panel as set forth in Section 10.5.7.

10.5.6. Drainage Review Panel ... In order to review the

need for Drainage Pumping, if the Chief Executive Officers fail to agree, initially and subsequently, if requested, City and DEW together shall appoint one member of a Drainage Review Panel and District shall appoint one member of a Drainage Review Panel and the two panelists so appointed shall appoint a third panelist. If any panelist is not appointed within thirty (30) days of a request by any Party to do so, then any Party may request the American Arbitration Association to appoint the panelist. Each panelist shall be a person with extensive experience in groundwater management matters and shall also be either a California registered geologist or a California registered civil engineer.

10.5.7. Determination by Drainage Review Panel. Within six (6) months of its appointment the Panel so appointed shall determine whether the operation of the Project has resulted in a need for Drainage Pumping in order to protect the viability of crop production, including orchard crops, and incorder to prevent interference with structures or utilities in any portion of the District outside the Service Areas of City and DEW and specifically if, and the extent to which, the pumping is necessitated as a result of rising water tables resulting from the Project. The Drainage Review Panel shall, among other things, consider the affected area or areas, if any, the appropriate base or comparison period, and specifically whether necessary Drainage Pumping is resulting from the Project outside the Service Areas; of City and DEW. Finally the Drainage Review Panel shall determine the amount, if any, of Drainage Pumping made necessary as a result of the Project and anticipated to be made necessary during the next

7 12 mg

الأول المحافظ الحراب المحافظ العالم المحافظ ا

five years.

10.5.8. Findings and Decision. Within six (6) months of its appointment the Drainage Review Panel shall by a unanimous or majority vote render its findings. If at least two panelists cannot agree then the determination of the third panelist appointed by the two initial panelists shall be the binding determination. If Drainage Pumping is determined to be necessary the costs of the needed facilities including but not limited to. capital costs, depreciation of existing facilities and the cost of operations, shall be added to the costs of the Project to be borne by City and DEW in the year following the determination and each year thereafter until modified as provided in Section 10.5.9. TIT the Drainage Review Panel based on the previous five years (except the first five years) determines that Drainage Pumping made necessary by the Project has not been included as a Project cost during the preceding five years, or if the Drainage Review Panel determines that the Drainage Pumping cost has exceeded the actual cost of Drainage Pumping necessitated by the Project during the previous five years, then the Drainage Review Panel shall make an appropriate cost adjustment for the following five years. Conversely, if the Drainage Review Panel determines that during the previous five years that an amount has been included as a Project cost in excess of the actual cost of the Drainage Pumping necessitated by the Project during the last five years, then the Drainage Review Panel shall make an appropriate cost adjustment for the following five years. The cost of the Drainage Review Panel, including reasonable professional and

:: · ...

clerical fees shall also be a Project cost.

of the fourth (4th) year following a determination pursuant to section 10.5.8. and at five (5) year intervals thereafter, any Party may institute a further review of the continued need for Drainage Pumping and the amount thereof as a part of the Project which review shall be instituted and undertaken in the same manner as the initial review. A Drainage Review Panel shall be selected which may, but need not include members who served on a previous Panel.

- District shall not be responsible for any Drainage Pumping necessary to maintain water tables within the respective Service Areas of City and DEW.
- If District markets Drainage Pumping water as defined in Section 4.6 outside the District, the net proceeds, if any, attributable to the marketing of water resulting from Drainage Pumping shall be credited to the Project.
- lation of net proceeds attributable to the marketing of water resulting from Drainage Pumping is and will be difficult at the time of entering into this Memorandum and the Treatment and Delivery Agreement without reference to specific, presently unknown future drainage, water supply, and water use conditions.
 - 10.7.2. Review by Chief Executive Officers In the event that District markets water outside the District during a

والمراجع المراجع المعارض

period when Drainage Pumping is occurring, the District shall notify the other parties and the Chief Executive Officers of the Parties shall promptly meet and endeavor to agree upon the determinations to be made by the Water Review Panel pursuant to Section 10.7.4 and Section 10.7.5. If the Chief Executive Officers reach unanimous agreement, then their determinations shall apply as if made by the Water Review Panel as set forth in Section 10.7.4.

10.7.3. Water Review Panel. In order to establish the a the proposition sum, if any to be credited for water marketed outside the District while Drainage Pumping is underway, if the Chief Executive Officers fail to agree, initially and subsequently, if requested by any Party, City and DEW together shall appoint one member of a Water Review Panel and District shall appoint one member of a on had dam Od by loung whom to bring Water Review Panel and the two panelists so appointed shall The William Street Commence of the street of appoint a third panelist. If any panelist is not appointed ముందు మంది ఉద్యమ్ ముందిన స్థానికి ఉద్దారి ముందు ఉంది. within thirty (30) days of a request by any Party to do so, then any Party may request the American Arbitration Association to appoint the panelist. Each panelist shall be a person with త్ 1.2 మీ విజా - కుర్యాక్స్ అంది 2 అంటింది. మీ కార్యాక్స్ extensive experience in drainage and water management matters and shall also be a California registered civil engineer.

10.7.4. Determination by Water Review Panel. Within six (6) months of its appointment, the Water Review Panel so appointed shall determine whether Drainage Pumping has generated water marketed outside the District and if so, the amount of net income fairly attributable to water resulting from Drainage Pumping as opposed to water resulting from other sources.

....

n vin i Martin nes i li li gamini. Il nesti i sino i ni gali i ne si li li gama.

10.7.5. Findings and Decision of Water Review Panel. TO THE SECOND PRODUCT OF THE PROPERTY OF THE P Within six (6) months of its appointment, the Water Review Panel วิธี 25070 อีลต**์ อ**็ตลม แกรมสมอังการอดิสอ อสต หรืนสอส shall by a unanimous or majority vote render its findings. If at least two panelists cannot agree, then the determination of the ుగ్రామంలో అయ్య కూడా ఎక్కువ గ్రామం మూలు ప్రభుత్వముడు గ్రామం third panelist appointed by the two initial panelists shall be the binding determination. The determination of the Water Review Panel as to whether a credit should be allowed, the calculation ్తారు. ఇంద్రి మండుకులో అనింది అని అవవుడు ఓమ్ ఉంది. ఇందిందు కెవ్విట్స్ of a credit, and the amount of the credit shall remain in effect until any Party asks for a further review in which case an <u>. M</u>r. 7<u>18.</u> Tojak 16. 18.7.01 additional review shall proceed as set forth above for the That laife is an element whose was promised in the second in the initial review. చికా ఇక్కారాడు మందుకు కుమ్. ద్వామిక కామాక్ క్రామించికుండాడు. అనినిటింగా జీతా కామాక్

11. DELIVERY OF WATER.

11.1. Water Supply to City and DEW. Subject to Sections
4.17 and 11.2, District shall make available to City and DEW an
amount of water equal to 30 million gallons per day. District
shall consult with City and DEW on a regular basis during the
Commercial Operation Phase to determine the schedule of deliveries, and, consistent with the terms of this Memorandum and the
Treatment and Delivery Agreement, shall use its best efforts to
meet the requirements of City and DEW.

11.2. Formula for Water Entitlement. The maximum amount of water from the Project to be made available by the District annually to City and DEW, collectively, during the Commercial Operation Phase shall be determined by a formula set forth in the Treatment and Delivery Agreement which will relate the annual entitlement of City and DEW to total water available to District, and will implement the provisions of Section 11.5. The formula

Committee Income process.

اً الوقيد من أي واقر يوم يستوال الحربي يوم المجال المجال المستوال المستوار المستوار المستوار المستوار المجال م المجال المراجع معالم المجال المجال المستوار المراجع المجال المجال المجال المجال المستوار المستوار المجال المجا shall establish the maximum amount which District is required to deliver to City and DEW each year. Subject to Section 11.5.1, District may, in its sole discretion, deliver to City and DEW The state of the s more water than the maximum amount determined by the formula calculation applicable in any year. In no event shall the District be required to make available more than 33,602.1 acre feet in any twelve-month period, and provided that if the applicable formula during any period provides for an annual entitlement of less than 33,602.1 acre feet there shall be no suspension CONTRACTOR OF THE STORES in the payment obligations of City and DEW regardless of the ్రవార్లు కారులోని కామ్మార్క్ జర్గు అందినానికి ఉంది. length of the period and further provided that nothing contained in this Section 11.2 shall be deemed to modify in any way the ' The second of th District's right to suspend, curtail or reduce water deliveries as provided in Section 8.7 and in the first sentence of this Section of this Memorandum. Nothing in this Memorandum shall be والعالم ويواكس والمواجع والأراب والأسارة المراجع والمراجع والمارات المراجع والمارات والمراجع والمارات construed to require District to curtail deliveries of water in The Control of the Co any year.

11.3. Level Deliveries. Unless the Parties agree otherwise, the 33,602.1 acre feet annual entitlement shall be delivered at ් බ්ට්ට් සම්බන්ධ වෙන්නී වන සහ මෙය කණුමක්කුව විනුදුකුණ the rate of 30 million gallons per day. In the event there is an రాజాబాద్నాలు కథ క్రామం ఉంది. మీద్రమైడ్ను annual entitlement of less than 33,602.1 acre feet pursuant to the formula in the Treatment and Delivery Agreement then the lesser amount shall be divided by 365 and the result shall be కు గ్రామంలో ఎంత్ కార్యాన్నారి. మూడ్కు కార్మాన్ క delivered on an equal daily basis unless the Parties agree ్ట్ కుర్వమ్మిక్ కున్నా ఉంది. ఎట్ ఎమ్. కమ్, కూడా ఎన్నా కెస్ట్ otherwise. In the event that District first determines that the The second of the control of the con water is not needed by agriculture, nothing in Section 11.2 and อาทิสต เหมอดหวอ โดยเพื่อชาวเกิดสารสารสารส this Section 11.3 shall be deemed to prohibit District from

- 2 A+

making additional quantities of water available, if either or both City and DEW are willing to accept the water.

- 11.4. Drought, Proportional Curtailment. The Treatment and Delivery Agreement shall address:
- 11.4.1. <u>Curtailment</u>. The curtailment of surface water deliveries in times of shortage when the District is unable to deliver 30 million gallons per day due to a drought situation.
- 11.4.2. <u>Conjunctive Use</u>. Related conjunctive use principles.

Table & American Confidence

- 11.5. <u>Drought</u>. The Treatment and Delivery Agreement shall also define and clarify what constitutes a drought situation when District may reduce the 30 million gallons per day deliveries, and shall set forth the precise formula for curtailment consistent with the following basic principles:
- delivery of surface water by District for agricultural uses to its agricultural customers as compared with the delivery for municipal uses to City and DEW shall be of equal priority, with deliveries for agricultural and municipal uses being curtailed in equal proportions. The curtailment of surface water shall not affect the production of groundwater for agricultural or municipal uses.
- 11.5.2. Adjustment of Curtailment. In the event that water deliveries must be curtailed pursuant to Section 11.5.1, Section 11.2, or the Treatment and Delivery Agreement, the parties shall meet and confer for the purpose of reaching an agreement as to an alternative curtailment formula or water

and the second of the second o

allocation basis which more equitably and more fairly meets the then current needs of the agricultural and domestic water users within the District's boundaries. The parties shall also endeavor or to reach agreement upon other terms and conditions necessary to implement such an agreement. The duration of such an adjustment agreement shall be specified in the agreement. In the event that no such agreement can be reached, the deliveries shall be curtailed in accordance with the provisions of Section 11.5.1,

situation, City and DEW shall continue to have the option to deliver groundwater to the irrigation canal system in exchange for additional surface water as provided by Section 8.6., subject to the standards set forth in Section 8.6.1.

Treated Water to Remain in Irrigation District Boundary. No
Treated Water delivered by District to City or DEW shall be
allowed to flow outside the District's Irrigation District
Boundary (as opposed to its Electric Service Area Boundary). In
order to ensure that City and DEW comply herewith, at no time
shall the delivery of water by either City or DEW within
District's Irrigation District Boundary be less than the amount
of Treated Water delivered at the time to City or DEW, as the
case may be, by District. In order to demonstrate compliance
with this Section 11.6, City and DEW shall maintain meters at
appropriate locations on their respective water delivery systems
to record (a) the amount of all water pumped from the groundwater
within the Irrigation District Boundary of District, and (b) the

- 0 C +

amount of all water transported outside the Irrigation District
Boundary. City and DEW shall monitor these records to ensure
that the amount recorded under (a) is at all times greater than
or equal to the amount recorded under (b). The records maintained by City and DEW under this Section shall be open to
reasonable inspection by District, and its agents, employees, and
officers. A monthly summary of the applicable records shall be
provided to District by City and DEW.

12. PROJECT MANAGEMENT.

12.1. District's Board of Directors.

forth in Section 12.3 and Section 12.5 and subject to the review set forth in Section 12.5 and Section 12.6, District's Board of Directors shall be the final decision making authority with regard to the Project. The Board of Directors shall consider the recommendations of the Project Manager. All decisions shall be made at duly noticed regular meetings or special meetings.

12.1.2. Project Manager. District's Board of Directors shall appoint a Project Manager, who will perform those duties set forth in Section 12.2.

12.2. Duties of Project Manager.

12.2.1. Implementation of Memorandum. The responsibility for implementing and administering this Memorandum and for carrying out the tasks necessary for the successful completion of each of the phases of the Project shall be vested in the Project Manager.

12.2.2. Notice to Advisory Committees. The Project

Manager, in the course of fulfilling his responsibilities, shall the party of the vocation of the provide the Advisory Committees reasonable advance notice of the impending major decisions as defined in Section 12.6.1.C.

The first of the first of the first war and the first of the first of

12.2.3. Specific Duties. The Project Manager shall have the following powers, duties, and responsibilities:

12.2.3.A. <u>Work Plans</u>. Developing a work plan for each Phase of the Project.

12.2.3.B. <u>Progress Reports</u>. Submitting periodic progress reports to the Parties and to District's Board of Directors.

12.2.3.C. <u>Administering Contracts</u>. Administering the contracts as may be entered into pursuant to the Treatmont and Delivery Agreement.

12.2.3.D. <u>Public Information</u>. Serving as public information officer.

12.2.3.E. Recommending Consultants. Recommending the hiring or utilization of consultants, engineers, contractors, attorneys, underwriters, and other services necessary to carry out the Project.

12.2.3.F. Financial Report in Preparing and annual in Elnancial report of the operation of the Project for reviewaby the Advisory Committees in a committee of the committees in the Advisory Committees in the committee in the committees in the committees in the committee in the committees in the committee in the co

12.2.3.G. Other Duties. Other duties as may be necessary and proper to carry out the Projections 12.2.3.

12.3. City and DEW Approval of Project Milestones. Since City and DEW will be paying virtually all costs associated with the design, construction, operation, and maintenance of the

Project, the Parties agree and consent that City and DEW shall have the authority to approve or disapprove major Project mile-

- 12.3.1. Engineering Services. Prior to District's approval of any engineering services contract concerning the design and engineering of the Project, City and DEW shall review the contract or contracts and each shall provide District with its written authorization to proceed.
- 12.3.2. Review of Preliminary Design. Prior to proceeding with the final design phase under an engineering services contract, City and DEW shall review the preliminary design plans and preliminary cost estimate and each shall provide District with its written authorization to proceed.
- 12.3.3. Acceptance of Final Design. Prior to accepting the final design work, including construction ready final plans and specifications and final cost estimate, City and DEW shall review the final design work and cost estimate and each shall provide the District with its written authorization to proceed.
- 12.3.4. Construction Contracts. Prior to District's approval of a construction contract or contracts for the construction of the Project, City and DEW shall review the construction contract documents and each will provide District with its written authorization to proceed.
- each act promptly in giving or refusing to give any of the written authorizations to proceed set forth above. The written authorizations to proceed to be given by City and DEW shall not

Same and the same of

Harris Committee Committee

be unreasonably withheld. If City or DEW fails to provide its written authorization to proceed, or written refusal of authorization, within forty-five (45) days from the date of the written request for authorization from District, then City or DEW shall ್ ಬಳ ಪ್ರಾಥಾಣಕ್ಕೆ be deemed to have authorized District to proceed with the mile-The state of the s stone in question. If City or DEW refuses to authorize any of 139 NAPL 255 23 the foregoing Project milestones, the Party refusing shall set المن الدوال المداد المراد الالداد المراد الدوال المناد المراد المراد المناد المراد المناد المناد المناد المناد forth in writing its reason or reasons for the refusal and shall timely provide the writing to the other two Parties. Thereafter (a) District may work to address and resolve the Party's concerns the state of the s and then re-request the written authorization to proceed for the particular Project milestone, or (b) any Party may seek to Marie San State State Control terminate this Memorandum pursuant to Section 18.2.

- 12.5. Project Financing Approvals. Prior to issuing bonds, certificates of participation, or other evidences of indebtedness for the purpose of financing the Project, District shall give written notice to City and DEW that the time to commence financ-A TIMESERAL ABOUT OF SOME ing has arrived. Within thirty (30) days of the notice, the Finance Committee shall meet and confer at least once and within sixty (60) days of the notice City and DEW shall advise District in writing as to whether Fixed Financing or Variable Financing shall be used. If either or both City and DEW advise in writing - United to the second of the that Fixed Financing is to be used or fail to give their written advice within sixty (60) days of the notice, then Fixed Financing ស៊ីជ័ព ស៊ីមា ១០ ១ ១០ ៦៤១ ស្រីមា ជាស្ថិតក្នុង២០២៤ម្តី shall be used. THE STATE OF THE S
 - 12.5.1. Fixed Financing. If Fixed Financing is to be used, District shall give written notice to City and DEW that

.

A Commence of the Commence of

Fixed Financing is to be used and shall request that City and DEW each give to District a not to exceed TIC within thirty (30) days of the notice and request. Within fifteen (15) days of the notice and request by District, the Finance Committee shall meet at least once. Upon receipt of a TIC from both City and DEW, District shall thereafter be responsible for marketing the financing subject to the lowest not to exceed TIC approved by .City or DEW. The members of the Finance Committee shall be present at the time and place of marketing and they and their 774 10 0 0 0 7 3 mg N financial advisers shall consult with the persons responsible for the marketing for District and the terms and conditions of the financing, but all final decisions shall be made by District, ' provided, that the TIC of the issue is (a) the lowest then available, and (b) below the not to exceed TIC approved by City and DEW. In the event that either or both City and DEW fails to approve a not to exceed TIC within thirty (30) days of the notice from District that Fixed Financing is to be used then the Parties shall seek to reach agreement as to a not to exceed TIC or any Party may seek to terminate this Memorandum pursuant to Section on the American South State (Section 1997) Province Common Homestay & 18.2.

12.5.2. Variable Financing. Variable Financing shall be used only during the construction period, except as otherwise provided herein. If City and DEW both agree that Variable financing should be undertaken, the District shall issue Variable Financing for the duration of the construction period. No later than sixty days after the Commercial Operation date, as that term is defined in the Treatment and Delivery Agreement, the Variable

-75-

. Partitorial attendance of the

Financing shall be replaced entirely by Fixed Financing unless the Parties unanimously agree to the contrary.

shall initially be issued in such maturities and at such interest rates as, in the judgment of the District after consulting with the Finance Committee, provides the most cost-effective combination of such maturities and interest rates. Thereafter, as the securities mature, the District shall be solely responsible for the remarketing of such securities until the issuance of the Fixed Financing.

12.5.2.B. Conversion During Construction Period.

If Variable Financing is issued for the construction period,
either City or DEW may, during such construction period, request
that the District convert such Variable Financing to Fixed
Financing. Such request may be conditioned upon the ability of
the District to secure such Fixed Financing at a TIC specified in
the request.

12.5.2.C. Conversion Upon Commercial Operation.

If no request to convert Variable Financing to Fixed Financing has been made prior to the Commercial Operation of the Project, the District shall convert the Variable Financing to Fixed Financing on the Commercial Operation date, or within sixty days thereafter without regard to the provisions of Section 12.5.1, provided, however, that the District shall consult with the Finance Committee during the time it is converting the Variable Financing to Fixed Financing unless the Parties unanimously agree to the contrary.

- 12.6. Advisory Committees. In order to assist the District and the Project Manager with the implementation of the Project, the following committees are formed and shall have the following duties, responsibilities and authorities.
- formed consisting of two City Council members, two members of the District Board of Directors, and two officers of DEW. The Policy Committee shall meet at least twice each year, and at other times when a meeting is called by the Project Manager. Each member of the Policy Committee shall serve at the pleasure of the Party selecting that member. The Policy Committee shall function during all three phases of the Project, and shall have the following responsibilities and authority:
- recommendations to City and DEW concerning the request for authorization to proceed with Project milestones pursuant to Section 12.3. District and the Project Manager shall not request written authorization to proceed with any of the Project milestones until after the Policy Committee has reviewed the proposal and made a recommendation to City and DEW.
- 12.6.1.B. <u>Budget Functions</u>. To perform the functions with respect to the District budget for the Project as set forth in Section 12.7. All expenses of District concerning the Project shall be consistent with the approved budget.
- 12.6.1.C. Major Design Changes. To review and advise District and the Project Manager concerning major changes in the design of the Project. A major change in the design of

the Project shall be defined as any change involving an estimated increase or decrease in the cost of the Project in excess of \$100,000.

\$100,000). To review and advise District and the Project Manager concerning any major decision with respect to the Project. A major decision shall be defined as any contract, change order, purchase, change in policy, or any other action with an estimated cost in excess of \$100,000.

shall be formed, consisting of one staff person appointed by each Party, and one alternate member as each Party deems necessary.

Each member of the Technical Committee shall serve at the pleasure of the Party selecting that member. The Technical Committee shall meet at least once each month, and at all other times as requested by the Project Manager. The Technical Committee shall function during all three phases of the Project, and shall have the following responsibilities and authority:

advise and consult with the Project Manager and the District, to exchange information, and to make any necessary recommendations relating to the Project design, construction, operation, and maintenance.

review and make recommendations to the Policy Committee for all matters within the scope of authority and responsibility of the Policy Committee;

and the second

APPENDIX A Page 44

12.6.2.C. Advice Relative to Minor Decisions

(\$5,000.00 to \$100,000.00). To review and advise the Project Manager concerning any minor decision affecting the Project. A minor decision shall be defined as any contract, change order, purchase, change in policy, or any other action with an estimated cost in excess \$5,000, but not in excess of \$100,000, and any action involving changes in the Treated Water quality beyond the range of normal plant operation variability.

12.7. Budget.

12.7.1. Budget Principles. In preparing and reviewing budgets for the Project, the Parties shall be guided by the principle that the Project shall be operated in as economic a manner as practical in accordance with generally accepted waterworks practices as evidenced by similar sized clean water treatment plants in Northern California.

12.7.2. Budget Process.

Commercial Operation Phase (Phase III) of the Project shall be prepared by the District during Phase II, allowing ample time for review by the Technical Committee and the Policy Committee, prior to the commencement of Phase III. For each subsequent fiscal year during Phase III, the District staff shall prepare a budget for the Project prior to the beginning of each fiscal year. All Project budgets shall include both operating and capital components.

mendation. Prior to completion of the first administrative draft

and a lead one of the page .45.

of each year's budget by the District Staff, the Project Manager shall meet with the Technical Committee at least once to discuss and receive input from the Committee concerning development of the budget. Prior to submission of the budget to the Policy Committee the Technical Committee may, if it elects to do so, make a recommendation with respect to the budget to the Policy Committee. Upon completion of the first administrative draft of a year's budget, the Project Manager shall forthwith furnish it to the members of the Technical Committee. Within 15 days of receipt of the draft budget, the Technical Committee members may individually or collectively submit to the Project Manager their recommendations and/or comments regarding the draft budget. The documents shall not be mailed to the City and DEW as provided in Section 12.7.2.C until after the expiration of this 15 day The state of the s period.

12.7.2.C. Information to City and DEW. At least 20 days before the submission of each annual budget to the District Board, the Project Manager shall mail to the City and DEW copies of the proposed budget for the ensuing fiscal year and a detailed calculation of the proposed payment obligations of the City and DEW calculated pursuant to Section 10.3 and the Treatment and Delivery Agreement. Commencing after the first year of operation, the Project Manager shall also at the same time mail copies of the most recent update of the current year's expenditures and revenues, and the balance sheet and income statement for the most recent fiscal year, if available.

12.7.2.D. Policy Committee Review. Approximately

Board, the Policy Committee shall meet to review and discuss the proposed budget for the ensuing fiscal year. The District staff at this meeting shall explain and justify the need for all of the various budget items and proposed expenditures. At this meeting, the Parties shall strive to agree upon a budget for the ensuing fiscal year.

Committee. The budget shall be deemed approved by the Policy
Committee unless, at the meeting, the Committee by an affirmative

vote of at least four members of the Committee objects to one or more of the budget items. If the Policy Committee objects to one

or more budget items, the Committee members objecting shall specify the item or items of the budget which are objectionable,

and why the item or items are objectionable.

12.7.2.F. Revision of Budget, Alternate Budget and

Report. If one or more budget items are objected to by the Policy Committee, the District staff shall either (a) revise the budget at the Policy Committee meeting so that it is acceptable to at least three members of the Policy Committee, (b) continue the Policy Committee meeting and thereafter consider the objections raised at the meeting and prepare a revised budget for consideration by the Policy Committee at a subsequent meeting, or (c) the District staff may determine to submit the budget to the District Board over the objections of the Policy Committee. If the District staff determines to submit the proposed budget to the District Board over the objection of four members of the

o Sym

Policy Committee the District staff shall give written notice to the Policy Committee and the Policy Committee may within 10 days after receipt of this notice submit an alternative to the objectionable budget item or items to the District Board to be considered along with the budget submitted by the District staff. The alternative budget item or items shall be accompanied by a report as to the reasons the alternate budget should be adopted in place of the budget submitted by the District staff.

- 12.7.3. Public Hearing. The annual budget for the Project shall be adopted by the Board only after a public hearing for which notice has been given in the same manner as currently provided in Section 21378 of the Water Code.
- 12.7.4. Appearance at Hearing. The foregoing provisions on review concerning the budget are not intended to, and shall not, preclude the City and DEW, and their respective officers, employees and agents, from appearing before the District Board regarding the proposed budget.
- 12.7.5. Augmentation. If, during the course of any fiscal year, the District proposes to make any budget augmentation, then the provisions of Sections 12.7.1 through 12.7.2.F concerning review of budgets shall apply to the budget augmentation.
 - 13. TREATMENT AND DELIVERY AGREEMENT.

The second secon

13.1. Negotiation. Upon execution of this Memorandum, the Parties shall begin to negotiate in good faith and to proceed with diligence to arrive at the necessary terms and conditions to enter into and execute a Treatment and Delivery Agreement, with

in the contract of

the objective of having said Treatment and Delivery Agreement effective no later than the start of Phase II.

- 13.2. Phases II and III. The Treatment and Delivery Agreement will define the working relationships between the Parties during Phase II and Phase III.
- 13.3. Accounting and Arbitration. Among other things, the Treatment and Delivery Agreement will also specify the treatment of depreciation, overhead, and reserves, if any, and will provide for the arbitration of specified disputes. The Treatment and Delivery Agreement shall also provide that net investment earnings from the proceeds of the issuance of bonds, certificates of participation, or other evidences of indebtedness shall be Transport of property applied to costs of the Project.
- 13.4. Further Detail in Treatment and Delivery Agreement. It is the intention of the Parties that to the extent necessary, the general provisions of this Memorandum shall be further defined and specified in the Treatment and Delivery Agreement which shall provide for the implementation of the concepts set forth in this Memorandum. ారు. ఈ కాణ ఈ సందర్భామార్గా అధ్యాహిక్షింది. రోహి స్థానికాస్తో చేశుకారికిపి

14. LIABILITY.

14.1. Insurance, General. During all phases of the Project, District shall maintain Commercial General Liability insurance or Comprehensive General Liability insurance, Automobile Liability insurance, Umbrella or Excess Liability insurance, Property insurance, and Workers Compensation insurance for the Project, including the construction, operation, and maintenance of the Project, and all operations and activities concerning all phases ್ ೧೯ ಕರ್ಮ<mark>ವಾರ</mark>ವರಲ್ಲಿ ಅಂತರ್ವಾಗಿ ಹಾಗೂ <mark>ಶರ್ವಾ ಇದ್ದಾರ್</mark>ಗ

of the Project. City and DEW shall be additional insureds on all policies required by this Section 14. The District's insurance coverage shall be primary insurance coverage as respects the City and DEW. Any insurance or self insurance maintained by the City and DEW shall be excess of the District's insurance and shall not contribute with it.

General Liability Insurance. Commercial General Liability or Comprehensive General Liability Insurance shall include coverage for Premises and Operations, Contractual Liability, Personal Injury Liability, Products/Completed Operations Liability, Broad Form Property Damage and Independent Contractors Liability, in an amount not less than one million dollars (\$1,000,000) per occurrence, combined single limit, two million dollars (\$2,000,000) aggregate.

14.1.2. Comprehensive Automobile Liability Insurance.

Comprehensive Automobile Liability insurance including, as applicable, owned, non-owned and hired automobiles, in an amount not less than one million dollars (\$1,000,000) per occurrence combined single limit.

14.1.3. Umbrella or Excess Liability Insurance.

Umbrella or Excess Liability insurance in an amount not less than ten million dollars (\$10,000,000) over and above the underlying limits with the Umbrella or Excess Liability policy containing insuring agreements, exclusions and conditions of coverage substantially similar to the underlying policies.

14.1.4. Workers Compensation Insurance. Workers

grade and the second

Compensation insurance shall be written as required by the State of California, including employer's liability limits of not less than one million dollars (\$1,000,000) per accident.

- shall cover all buildings and structures comprising the Project, and all fixtures, equipment and facilities located in, on, or connected with the Project, and shall be written on a Replacement. Cost Basis.
- 14.1.6. <u>Insurance Provisions</u>. Each insurance policy required by this Agreement shall contain the following clauses:
- 14.1.6.A. Cancellation. This insurance shall not be cancelled, limited in scope or coverage or non-renewed until after thirty (30) days prior written notice has been given to District, City, and DEW.
- 14.1.6.B. <u>District and others to be Named</u>. The District, its directors, officers, agents, employees, representatives, and volunteers are named as insureds, except Workers Compensation insurance.
- 14.1.7. Approval of Insurers. The policies of insurance required under this Memorandum shall be issued by an insurer or insurers, and shall be in a form, approved by City and DEW.
- and all contracts entered into concerning the construction, operation or maintenance of the Project, District shall include a provision in any and all contracts requiring that the contractor provide insurance protection similar to or in excess of the insurance coverage set forth above.

~ <u>0</u> 5 -

- 14.2. Indemnification and Hold Harmless. Except as provided in Sections 14.3 through 14.6, District shall indemnify, defend, protect, and hold harmless City and DEW, and their respective officers, employees, and agents, from any and all liabilities, claims, damages, losses, judgments, penalties, costs, or expenses (including attorneys' fees) arising from or in any way connected with the construction, operation or maintenance of the Project by District or by District's employees, contractors, subcontractors, consultants, or agents, or any one employed directly or indirectly by any of the foregoing, regardless of any negligence on the part of City or DEW or any of their officers, employees or agents.
 - 14.3. Third Party Claims, Not Covered by Insurance, Willful. With respect to claims and lawsuits against one or more of the Parties by third parties concerning injury, death, or property damage resulting from the construction, operation or maintenance of the Project, which claims and lawsuits are not covered by insurance, including self insurance, maintained by District pursuant to Section 14.1, and which are the result of willful 70-2 Ad 0708 misconduct, intentional tort, or gross negligence of one of the Parties, the Party whose willful misconduct, intentional tort, or gross negligence resulted in the damage claimed by the third party shall indemnify, defend, protect, and hold harmless the other two Parties, and their respective officers, employees and agents, from any and all liabilities, claims, damages, losses, judgments, penalties, costs, or expenses (including attorneys' fees) resulting from a claim or lawsuit by a third party.

- 14.4. Third Party Claims, Not Covered by Insurance, Ordinary. With respect to claims and lawsuits against one or more of the Parties by third parties concerning injury, death, or property damage resulting from the construction, operation or maintenance of the Project, which claims and lawsuits are not covered by insurance, including self insurance, maintained by District pursuant to Section 14.1, and which are not the result of willful The second secon misconduct, intentional tort, or gross negligence of one of the Parties, District shall defend the claim or lawsuit on behalf of all Parties to this Memorandum which are named in the claim or lawsuit, and District shall pay any settlement entered into by District or judgment entered against District or a Party. City and DEW shall reimburse District for its defense costs (including attorneys' fees and litigation expenses), settlement and judgment amounts incurred pursuant to this provision, in accordance with their respective percentage obligations to reimburse all Project costs pursuant to this Memorandum or a subsequent Treatment and ను కారాలు కూడు కుండు కావించిన ఉంది. చిందిన చేసిన చేసి Delivery Agreement. The state of the state of
- 14.5. Claims Between Parties. With respect to claims and lawsuits by one or more of the Parties against one or more of the other Parties, the claims and lawsuits will be processed and resolved in accordance with (a) the Tort Claims Act (or general tort principles if the action is against DEW) and/or (b) breach of contract remedies provided by this Memorandum, a subsequent Treatment and Delivery Agreement, or applicable law. Nothing in this Memorandum shall relieve any Party of any contractual liability or duty imposed by this Memorandum or a subsequent

- C 1/2 100

and the state of t

Treatment and Delivery Agreement.

- 14.6. Workers' Compensation Claims. Each Party shall bear the costs of discharging all liability imposed, including costs and expenses for attorneys' fees and other costs of defending, settling, or otherwise administering claims arising out of workers' compensation or employers liability claims brought by its employees.
- 14.7. Replacement of Pro Rata Right of Contribution. The insurance, indemnification, hold harmless, and reimbursement provisions set forth above in Sections 14.1 and through 14.4 are intended to and shall replace, and be applicable instead of, the pro rata right of contribution provisions of Government Code Section 895.6, to the extent the Section is applicable.

15. RELATIONSHIP OF PARTIES.

aran unula 200 bra kilojojoć Except as provided in Section 14, the covenants, obliga-ఎన్నాష్క్రాగ్ **న**హ కారావర్యం tions, and liabilities of the Parties are intended to be several ుడకి కుండికి దబ్దువాయం క్రిక్స్తిడ్డు కథ **ద్ర**్ and not joint or collective, and nothing herein contained shall The Complete section was ever be construed to create an association, joint venture, trust, or partnership, or to impose a trust or partnership covenant, obligation, or liability on or with regard to any one or more of the Parties. Each Party shall be individually responsible for its own covenants, obligations, and liabilities pursuant to this Memorandum. No Party shall be under the control of or shall be deemed to control any other Party or the Parties as a group. No ్రం కొట్టు గాలికిస్తో ఉన్నాయికిందా Party shall be the agent of or have a right or power to bind any ్ సర్సినికి కు. తీవ్రా డితరికట్టుం other Party without its express written consent, except as a po 111 - Suringa ingakera j expressly provided in this Memorandum.

- P & L

.

16. ASSIGNMENT.

CONTROPINA STOR No Party shall assign or transfer its right or interest in this Project, in whole or in part, without the written consent of each of the other Parties, which consent shall not be unreasonably withheld. Any assignment or transfer by a Party without consent shall be void and shall result in a termination of that Party's right or interest. No assignee or transferee shall obtain any right or interest in this Memorandum until It assumes by written instrument all obligations under this Memorandum with respect to the right or interest transferred or assigned, and it becomes a Party to this Memorandum by executing this Memorandum or amendment thereto. Nothing in this Memorandum shall allow District to sell Treated Water from this Project to any person or entity other than City and DEW without the written consent of both City and DEW which consent shall not be unreasonably withheld. Prior to selling Treated Water from the Project to any person or entity other than the City or DEW, District shall first offer the water to City and DEW, in amounts based on each Party's relative share of the Project water and costs at that time, on the same terms as set forth in the then existing Treatment and Delivery Agreement. The price of any Treated Water sold by District to a third party shall be not less than the full cost of producing the water including capital, Sunk Costs, treatment, and administrative costs, and in any event shall not be lower than the price to City and DEW. In the event the cost of furnishing water to a third party is less than the charges to a third party, the excess revenue shall be applied to the operation of the

ana talah tang kentangan di kelanggan di talah taka

The state of the s

State of the state

And the second of the second o Project. Subject to the provisions of this Section, this Memorandum shall inure to the benefit of and be binding upon the Parties and their respective successors and assignees. Notwithstanding the foregoing provisions of this Section-16, DEW shall have the right without the prior written consent of the other-Parties, to assign or transfer its right or interest in the Project to an assignee or transferee of all, or substantially all, of its public utility assets or of all, or substantially all, of those DEW public utility assets located within District's Irrigation District Boundary, provided that concurrently with such assignment or transfer, such assignee or transferee shall assume the obligations to continue to operate the utility assets The control of the control so transferred and the retail water business attached thereto and provided, further, that the security required by Section 10.4 and Bollar Madeo (c الرابي والخروم فيحت فيعار والماك agreed to by District and DEW either remains in place or is The said of the said of the said replaced (a) by other security in, or substantially in, compara-The second secon ble form and substance, or (b) if such assignee or transferee المتعافرة المواثقة والمتاريخ shall be City, by inclusion of other security which satisfies the Kaling Kalabapatan (Mil rate covenant theretofore given District by City under said A COLOR TOWN FOR THE COMMENT OF THE PROPERTY OF LANCE TRANSPORT Section, 10.4. $x\in \mathcal{C}(x)$. The integral $x\in \underline{\mathcal{C}(x)}$, $x\in \mathcal{C}(x)$, we have $x\in \mathcal{C}(x)$. We have

17. GENERAL PROVISIONS GOVERNING MEMORANDUM.

17.1. Severance. In the event that any of the terms, covenants, or conditions of this Memorandum or the application of any term, covenant or condition shall be held invalid as to any Party or circumstance by any court having jurisdiction over the Parties or subject matter of this Memorandum, all other terms, covenants or conditions of the agreements and their application shall not

be affected thereby, but shall remain in force and effect unless a court holds that the provisions are not separable from all other provisions of this Memorandum.

- 17.2. Waiver. The waiver at any time by any Party of its rights with respect to a default or other matter arising in connection with this Memorandum shall not be deemed a waiver with respect to any subsequent default or matter.
- 17.3. Counterparts. This Memorandum may be executed in counterparts. Each Party shall deliver its executed counterpart to District, which shall deliver a fully conformed counterpart to each Party.
- 17.4. Supporting Resolutions and PUC Approval. Each Party represents that it has legal authority to enter into this Memoa house in the second of the second randum and to perform its obligations hereunder, and shall attach The transfer of the control of the control to this Memorandum a duly authorized resolution evidencing the ال المستقدم ا authority. The person executing this Memorandum on behalf of each Party shall also be authorized to do so by resolution of its en en Governing Body, which shall also be attached to this Memorandum. of the same of the Prior to execution of this Memorandum, DEW shall also obtain approval to execute this Memorandum from the California Public Utilities Commission and a copy of that approval shall be at-The second of the second of th tached to this Memorandum.
- 17.5. No Rights in Other Parties. This Memorandum is for the sole benefit of the Parties and shall not be construed as granting rights to any other person other than the Parties or, except as specifically set forth in this Memorandum, imputing to any person the obligations imposed on a Party.

- 25--

- written instrument duly executed by all of the Parties hereto.
- 17.7. Obligations Prior to Termination. The obligations of the Parties incurred pursuant to this Memorandum prior to termination of this Memorandum shall survive the termination.
- Memorandum are inserted merely to facilitate reference and shall have no bearing upon the interpretation of any of the terms and provisions hereof.
- execute; and deliver any and all documents reasonably required to implement this Memorandum.
- 17.10. Each Term Independent. Each term, covenant, and condition of this Memorandum is deemed to be an independent term, covenant, and condition, and the obligation of each Party to perform all of the terms, covenants, and conditions to be kept and performed by it is not dependent upon the performance by the other Party or Parties of any or all of the terms, covenants, and conditions to be kept and performed by the other Party.
- 17.11. Governing Law. This Memorandum shall be interpreted, governed by, and construed under the laws of the State of Cali-
- July 18. green and termination.

Santa Carlo Carlo Carlo Carlo

become effective until it has been executed by all Parties. At that time, this Memorandum shall continue in effect until the earlier of the following:

- 18.1.1. <u>Superseded by Other Agreement</u>. This Memorandum is superseded by a Treatment and Delivery Agreement or other agreement which, by its terms, supersedes this Memorandum.
- 18.1.2. <u>Termination by Unanimous Agreement</u>. Termination by unanimous agreement of the Parties.
- 18.2. Termination of Participation Prior to Financing. Notwithstanding any other provision in this Memorandum to the contrary, any Party, prior to the time District issues bonds or certificates of participation or other evidence of indebtedness for the purpose of financing the Project, upon not less than thirty (30) days written notice to the other two Parties, shall be entitled to terminate its participation in the Project at any time (a) the Party determines that the Project is not feasible because of technical, engineering, or economic reasons, or if adequate insurance is, or probably will, not be available at a commercially reasonable price, or for other reasons as would cause a reasonably prudent utility in the same or similar circumstances to terminate its participation in a project as is contemplated by this Memorandum, or (b) City or DEW determines that District has failed to meet its Phase I or, if financing has not yet been obtained, its Phase II obligations, and has failed to pursue the obligations with due diligence. If the Memorandum is terminated by DEW, then City at its sole option may maintain this Memorandum in full force and effect as between City and District by agreeing in writing to assume all of DEW's rights, entitlements, obligations and liabilities pursuant to this Memorandum. If the Memorandum is terminated by City, then DEW at its sole

option may maintain this Memorandum in full force and effect as between DEW and District by agreeing in writing to assume all of City's rights, entitlements, obligations and liabilities pursuant to this Memorandum.

19. UNDERTAKINGS.

public, under any provision of this Memorandum is rendered strictly as an accommodation and shall not constitute the dedication of the system or any portion thereof by the undertaking Party to the public or to another Party or any third party, and it is understood and agreed that any undertaking under any provisions of or resulting from this Memorandum by a Party shall cease upon the termination of the Party's obligations under this Memorandum or the Treatment and Delivery Agreement.

20. WATER RIGHTS AND OWNERSHIP.

Neither City nor DEW shall own or acquire any of District's water rights, but shall be entitled to the delivery of Treated Water in accordance with the terms of this Memorandum. Neither City nor DEW shall have any ownership rights in any of the facilities of the Project.

noidemorphs linesty 150 or 21. NOTICES.

Any notice, demand, or request provided for in this Memorandum shall be in writing, and shall be deemed properly served, given, or made if delivered in person or if sent by registered or certified mail, postage prepaid, to the persons specified below:

District:

Chief Executive Officer
Modesto Irrigation District
Post Office Box 4060
Modesto, CA 95352

and the second

A STATE OF THE STA

THE REPORT OF THE STATE OF THE

Page 60 City Manager City: - City of Modesto: which is any our want to any b Post Office Box 642 Modesto, CA 95353 Date of the Main territory President DEW: Del Este Water Company Post Office Box 3250 And Donnell Community Modesto, CA 95353 And the second s WMODESTO IRRIGATION CDISTRICT Control of the contro ానక్సర్క్ కానికి గ్రామంలో ఉంది. సినిమా కార్ సమీపట్ల తూరి మండుకున్న President $oldsymbol{arrho}$ a nowamog zana ma maaniya aka ka kak $oldsymbol{arrho}$ Attest: secretary of your of moreon of moreon of moreon Approved as to Legal Form: ್ಕೆ ೧೯೯೮ ಭಾಷವಳ ವಿ. ಇದೆ ಇಲ್ಲಾವಾಗು ೯೯೬ ಒದೆದೆ ಮರಣದಿ ಫ**CITYPOF MODESTO**ರಗಾರಿಸಿ ಪತಿ ಅರಹವು ైల్స్ ని కాకానికి కునాతకతోదకుకుండిని కాక్షాతించిని కునితి ఏతి గ్రామమ్మ <u>ರ್ಷಗಡಣಕ ಅಗ್ರೇ ಜಕನ್ಮೂ ರಕ್ಷಯ</u> . In mooned many lied brown and all Paul Baxter, Acting City Manager Noither Cary and Paul Come or and Libra WEE was gold worthin acos so le <mark>City Clerk</mark> es podesions od 11 mg, svo castpir zaspv Approved as to Legal Form: Stan T. Yamamoto, Eity Attorney - DEL ESTE WATER COMPANY a California corporation 21. NOTECTE. ಸಿಗ್ರಾ ಗಂಕಾಕರು, ಹೇಗುವಾಗ, ರಕ್ಷ ಹಾಗ್ಗಡಾಗಿಗಳ ವಿಶವಿಸುವಿರುವ ಸೆರಡ ಸಿನ ಜನಾಣ ಆಧಾನವರ್ಷನ್ನು (2) vans Alastoad nomino en llege des Presidentes ou albes mad and the mode to delivered in the democratic between the tree of the medical contractions of the modern of the mode

Approved as to Legal Form:

Here Secretary and model of the call because of them being the

The second secon 20020 42 Jamesear

Exhibit "A"

RESOLUTION NO. 90-50

DETERMINATION TO CARRY OUT THE DOMESTIC WATER PROJECT

and the second of the second o

WHEREAS, the Board of Directors of the Modesto Irrigation
District finds as follows:

Historically, the area in and around Modesto has received its municipal and industrial water supply from groundwater. In recent years, however, the area has experienced, and continues to experience, declining groundwater levels and a deterioration in the quality of groundwater. More strict water supply regulations and lower allowable limits for certain contaminants have been promulgated by state and federal authorities. The Modesto Irrigation District ("District"), the City of Modesto ("City"), and the Del Este Water Company ("Del Este") expect a continued decline in water quality and quantity unless an alternative source of water is developed.

In November, 1984, a report entitled 'Modesto-Ceres Area

Water Management Study', prepared by James M. Montgomery

Consulting Engineers, Inc., and Leedshill-Herkenhoff, Inc., (the

-1984 Montgomery Study') recommended a two-phase program for

conjunctive use of groundwater and treated surface water to meet

the water supply needs of the area.

In March, 1989, a report entitled "Modesto Surface Water Treatment Study", prepared by James M. Montgomery Consulting

Engineers, Inc., (the "1989 Montgomery Study") utilizing the findings of the 1984 Montgomery Study, made recommendations regarding the size and location of a water treatment plant and related transmission facilities, and suggested various alternative treatment plant sites to be considered. That study recommended that a site near Modesto Reservoir be the preferred alternative in the EIR prepared for the project.

Because the course of action recommended in the studies could have significant environmental effects, as defined by the California Environmental Quality Act ("CEQA") the District caused to be prepared an Environmental Impact Report ("EIR") to consider various project alternatives. The EIR was duly completed, and on April 10, 1990, at its regular board meeting, the District's Board of Directors certified the Final EIR has having been completed in compliance with the CEQA (Public Resources Code \$21000 et seq.) and the Guidelines for the Implementation of the California Environmental Quality Act (Title 14 C.C.R. \$15000 et seq.)

A Pilot Plant Study was conducted at the Modesto Reservoir site to determine the recommended process train for the proposed water treatment plant. The Pilot Plant Study concluded that direct filtration treatment, with pre-ozonation and chlorination for final disinfection would meet all-water quality and process performance criteria. The Pilot Plant Study was presented to and

المجاهدي المهم أنها ويعدنها في المحافظ المجاهدي المجاهدي المجاهدي والمجاهد المجاهد المجاهد المجاهد المجاهد الم المهم والمجاهد المجاهد المجاهد المجاهد في المجاهد المجاهد المجاهد المجاهد المجاهد المجاهد المجاهد المجاهد المج

ර එම වර්ග රාදුම් රජ්ය ගැන්න ජනයට මෙන් ගිනි සිටින් කිරීම කිරීම දෙන දෙන දුන්දුම් මේ සිත්තික් සිටින් සිටිම සිටින් මේ සිටින් සිටින් දෙන්න සිටින් සිට accepted by the District's Board of Directors on April 12, 1990.

The Board has duly reviewed and considered the information contained in the 1984 Montgomery Study, the 1989 Montgomery Study, the Pilot Plant Study, and the Final EIR.

Implementation of the project at the Modesto Reservoir site will have certain significant environmental impacts. The EIR identifies certain mitigation measures that can be implemented to reduce or eliminate each of the impacts. A list of each of the significant environmental impacts that are likely to occur, and the mitigation measures for each such impact, are set forth in Exhibit "A", attached hereto and incorporated herein by this reference.

Based upon the material presented in the Final EIR, including without limitation the Draft EIR, the comments and the responses thereto, as well as information presented to the Board at its meetings of April 10, 1990 and April 17, 1990, the Board finds that the mitigation measures set forth in Exhibit "A" will mitigate or avoid the significant environmental effects identified in the Final EIR. The Board also finds, based upon the same information, as well as the Pilot Plant Study and the information presented at its meeting of April 12, 1990, that the best alternative is the Modesto Reservoir site. A complete statement of findings with regard to the mitigation and avoidance of the significant environmental impacts is set forth in Exhibit "B", attached hereto and incorporated herein by this reference.

The California Environmental Quality Act requires the District to adopt a Mitigation Monitoring Plan when it determines to carry out a project which will have significant environmental effects. A Mitigation Monitoring Program has been prepared which will ensure that the mitigation measures identified in Exhibit "A" will be carried out. The Mitigation Monitoring Plan is set forth in Exhibit "C", attached hereto and incorporated herein by this reference.

Now, therefore, BE IT RESOLVED that the Board of Directors of the Modesto Irrigation District:

- 1. The Modesto Irrigation District shall carry out the Modesto Surface Water Treatment Project; the Project shall be located at Modesto Reservoir, and shall utilize the direct filtration process with pre-ozonation and chlorination for final disinfection;
- 2. The mitigation measures described in Exhibit "A" shall be incorporated into the Project;
 - 3. Such mitigation measures will eliminate or substantially lessen the significant environmental effects of the Project identified in the Final EIR;
- 4. The findings set forth in Exhibit "B" are hereby as as a same way as a same and a same a same and a same a
- 5. The Mitigation Monitoring Program set forth in Exhibit
 "C" is hereby adopted, and the District's Chief Executive Officer

ುಕ ಕಾರ್ಯಕ್ಷ ಕ್ರಮದ ಗಟ್ಟಿಸಿದ್ದಾರೆ. ಇತ್ತಿದ್ದ ಕರ್ಮಕ್ಷ ಕ್ರಮದಲ್ಲಿ ಮತ್ತು ಕ್ರಮದಲ್ಲಿ ಕ್ರಮಿಕ್ಸ್ ಕ್ರಮಕ್ಕೆ ಕ್ರಮಿಕ್ಸ್ ಕ್ರಮಿಕ

is hereby appointed Environmental Monitor for purposes of said Mitigation Monitoring Program; and Contractors.

A Notice of Determination shall be prepared and filed 6. in compliance with the requirements of the California Environmental Quality Act and the CEQA Guidelines: بالمراجع والانجيانية

Moved by Director Cowan, seconded by Director Billington, that the foregoing resolution be adopted.

Upon roll call the following vote was had:

Ayes:

Directors Lyons, Cowan, Beck, Billington, Kidd

Book Bargo is a number of supering of many

The first of the water water Caliena wayyee waxayee wax

REPART CORRESPOND CREE CORRECTION OF A CREEK CORRECTION OF THE CORRECTION OF THE CREEK CORRECTION OF THE CREEK CRE

BOA LANSON was the second of the second o

ومن منه و موروه و براي الراسيد و الراسيد

Carolian) ca

The Physical Co.

The map where the constant of the constant of

Noes:

Directors None

Absent:

A CONTRACTOR OF THE PROPERTY O

Directors None

The President declared the resolution adopted.

The energy of the second constraints of the COO NAT AND LOS LAND THE THE THE

I, VICKIE EHRLER, Secretary of the Board of Directors of the MODESTO IRRIGATION DISTRICT, do hereby CERTIFY that the foregoing is a full, true, and correct copy of a resolution duly adopted at a regular meeting of said Board of Directors held on the 17th day of April, 1990. දී රා දීමයට වී රජය මුල්ලවට කිරීම මාදරමේ

> Secretary of the Board of Directors of the Modesto Irrigation District

אים למינים לא ביינים היים לא לאים לאים היים לאים לא היים לא ביינים לא ביינים לא ביינים לא היים לא היים לא היים · "说是一个什么我们就要说:"我们,我们这种要是

THE SECTION OF THE SE framerican Inches Annes Ann I in Tarenamoiestell aculy A TOWN ALCHOUNT OF A

00 JUAN 192

35.000 J

33 176.3

APPENDIX A Page 66

SUMMA OF MED DOMESTIC WATER PROGRAM TSUNX COSTST

		77.5 	والمراجع والمحاجع والمحور والمراجع المراجع	An	nume of COSt	Shared by:	#ID
ID Board				TOTAL Amount	Crty of	Del Este Water Co.	"Zunk Cost"
	Date			and the major and and		44	
88-54	3/15/98	James N. Montgomery Development of a Water Tre Plant Siting Study, include - water requirements evalu	ingr		المحمد المحم المحمد المحمد		103.000.00
	• .	- plant capacity determine - mice evaluation; and	evaluation.		يوه الواقع والمنطقة المحمد		
	· · · · · · · · · · · · · · · · · · ·	of the grant of the second	and the second of the second o	76,000.00	173.600.00	³⁴ 74.400.00	248,000.00
87-53	5/ 1 /17	James M. Montgomery Authorizes completion of a study of alternative water processes and development preliminary design criter; of acudy to be used in the	pilot ctreatment for 5	g Marin Stage Communication (Communication Communication C	amen in Age amber of the grant	ரீ வை வடைது தில வட வடைந்துருக் ≹	
	*· · · · · · · · · · · · · · · · · · ·	Environmental Impact Repo	rt and to				
•		for the treatment plant.	in the second of	of the same	المراجع		
90-19	1/30/90	James H. Hontgomery Brief assessment of the p	د مانيون (۱۱ (کار خا <u>ن مړ و يون</u>	5,000.00	1,750.00		• 2.500.00
,,,,,	u, ,	information activities co	nducted to		2 7 d+ © 3		
		Program. Subcontractor:	20 Tells, and			·.	ت او الد
	SUBTOTAL	James M. Montgomery	,		947. 450 00cc	106,050.00	353.500.00
28-64		THE COMMUNICATES				13,764.61	
		Perform required environmental services and prepare required environmental documental the proposed municipal an industrial vater treatment	red regarding ic plant			,	
رین رستیه بیرسیاس سرساست در	or in the real of the same of	gurs Consultants		2 202 12	771.09	330,47	1,101.56
19-36	4/11/8	9 ORS CONSULTANTS, Addition	na I		پسريمي ميد ساز ددم جوه يخدن مادن معام داد دمانداد	4 (2) Ex (C) (2) (2) (2) (2)	
		studies, including conduc	eprare a mado	entrary of the second	The same of the sa		
***	اگر آموند. دور آموندارد	BEACTED' VOCTESTAL ALIES		به ب			5 944 S/
motion	اگر بهای در	o ors Consultants	iedrae do mado Opinio do mado Opinio do mado	به ب		7 3.1 7 1.763.35	5,944.50
motion	اگر بهای در	plant survey. Ours Consultants Items requested in URS letter of 6/22/89; - Site Analysis Report addendum, submitted to	9,164.00	11,889.00		a al 1707 1,753.35	5,944.56
rotion	اگر بهای در	plant survey. O URS Consultants Items requested in URS letter of 6/12/49: — Site Analysis Report addendum, submitted to MID 1/6/89. Additional EIR scoping	9,164.00 333 (1.54.00 333 (1.55) 335 (1.55)	11.889.00 j		a al por ₁₇ 6s	5,944.50
motion	اگر بهای در	plant survey. O URS Consultants Items requested in URS letter of \$/22/89. - Site Analysis Report addendum, submitted to	9,164.00 9,164.00 93012 930 20 752	11.889.00 j		a al Paraiss	5,944.50
motion	اگر بهای در	plant survey. O ORS Consultants Items requested in ORS letter of 6/21/89: Site Analysis Report addendum, submitted to MID 1/6/89. Additional EIR scoping - Additional EIR public hearings on project status. 2nd Review copy of Draft	000000 0000000000000000000000000000000	11.889.00 j		TA OLI	5,944.50
motion	2/20/9	plant survey. O URS Consultants Items requested in URS letter of 6/22/87: — Site Analysis Report addendum; submitted to —NID 1/6/89. — Additional FIR scoping — meetings. — Additional FIR public hearings on project status. 2nd Review copy of Draft EIR.	000000 2 2220 9.164.00 900120 0000 22 7520 21 22300000	11.889.00 j	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	of the mild (business of the second of the s	
motion 90-74	2/20/9	plant survey. O URS Consultants Items requested in URS letter of 6/22/87: - Site Analysis Report addendum; submitted to -NID 1/6/89. - Additional EIR scoping - Additional EIR public hearings on project status. 2nd Review copy of Draft EIR. 90 URS Consultants Amendment No. 2: Archae studies and additional w response to comments on as follows: - URS letter of 2/6/90	9,164.00 2,725.00 cological oct on pract fir,	11.889.00 j		of the mild (business of the second of the s	
	2/20/9	plant survey. O URS Consultants Items requested in URS letter of 6/12/99: — Site Analysis Report addendum, submitted to MID. 1/6/89. — Additional EIR scoping — Additional EIR public hearings on project status. 2nd Review copy of Draft EIR. 90 URS Consultants Amendment No. 2: Archae studies and additional w response to comments on as follows: — URS letter of 2/6/90 review of responses to Draft EIR. — URS letter of 4/30/90 a. cultural resources (archaeological) study	9,164.00 9,164.00 2,725.00 cological work on Draft EIR, 20,451.00	11.889.00 j	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	of the mild (business of the second of the s	
	2/20/9	plant survey. O URS Consultants Items requested in URS letter of 6/12/89: — Site Analysis Report addendum, submitted to MID 1/6/89. — Additional EIR scoping — Additional EIR public hearings on project status. 2nd Review copy of Draft EIR. 90 URS Consultants Amendment No. 2: Archae studies and additional w response to comments on as follows: — URS letter of 2/6/90 review of responses to Draft EIR. — URS letter of 4/30/90 a. cultural resources (archaeological) stud at three Alternative sites, and — b. final pristing, mailing and 50	9,164.00 9,164.00 2,725.00 cological work on Draft EIR, 20,451.00	11.889.00 j	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	of the mild (business of the second of the s	
	2/20/9	plant survey. O URS Consultants Items requested in URS letter of 6/2/47: — Site Analysis Report addendum, submitted to — NID 1/6/89. — Additional EIR submit hearings on project status. 2nd Review copy of Draft EIR. O URS Consultants Amendment No. 2: Archae studies and additional wresponse to comments on as follows: — URS letter of 2/6/90 - review of responses to Draft EIR. — URS letter of 4/30/90 a. cultural resources (archaeological) stud at three Alternative sites, and b. final printing,	9,164.00 9,164.00 2,725.00 cological ock on praft fix, 20,451.00	11.889.00 j	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	of the mild (business of the second of the s	14,761.0

APPENDIX A Page 67

SUMMAR MID DOMESTIC WATER PROGRAM -SUNK COSTS-

האגדבדבאר	CONTRAC	TS (CONE.)	erk in the property	ويد ويدر النوالية المدر ويد فيناه الدراء أن الدراء	रक्षाच्याच्या च्या का अनुसूक्ष्य कुल्या राज्य व्यक्ति जन्मजीवर वेलील	Amount	, of Cost	Shared by	HID
ID DOARG	Kesol.		en company	e ne speniero sum semestero	Total	2010 G17	y of lesto	Del Este Water Co.	"Sunk Cost"
90-74	6/19/70	AWC. IBC. Turbidity mosi - Two turbidit - Installation	toring sys	ალია ეგე ••••• ეგე	750.00 2000 750.00	. (() (() () () () () () () () () () () (,062.50	1,312.50	4,375.00
		The state of the s			10.500	.003	.062.50	1,312.50	6.125.00
	r vinni	_			(3) (3) 4,800				2,400.0
Pending	5/8/707	EIP Associates Site-specific	ponenical	studies		· echin	<u> چې چې چې د م</u>	عه وهو بدري و خد کار رادر مند	
6	Park of the Control	EIP Associates	5 9-40		m 33 <u>574,80</u> 0	0.00 Jan 2			2,400.0
* * * * * * * * * * * * * * * * * * * *	0.7199,700	and the property of the second	a to		857,65		,574.84	2428,389,123 <u>,</u> 2	429,686.1
OD LATOT	NSULTANT Ce: Inclu	CONTRACTS des installati	on by MID	oe \$1.750·.)	Land Contract Va	Ayra at		und danum	
		migration of the state of			e gyr en wer yn i yn yr en y yn yr ei y yn yr ei y yn yr ei y y y y y y y y y y y y y y y y y y				
KID DEDI	TT 41 VE	ATT TIME		-	e de des grants de communicación de servición de communicación de communic				rio.
Period C	overed	Staff Name	er og syk en og skalende er og en og skalende er og skalende er	1994.20	Total Amour	LE	odesco Stanot		Coef.
1/1-12/3	100832 11/88	Lee De Lano	(8 50%)		12 250 71.94	1230 mms	10 0:00	2	71.941. 60,909.
1/1-12/3	10/31/20 11/88 15/49/100	Lee De Lano Lee De Lano	(6 20r) (6 20r)	one o Neona Tarrag	400 71.94 60.90	12.20 mm a	10 0:00	a 7 2 4 2 70 - 00 - 1	71.941. 60,909.
1/1-12/3 1/1-10/3	1/11 13/19/25 15/19/25	Lee De Lano Lee De Lano	(@ 50%) (@ 50%)	<u>anda</u> Angara Angara Angara	200 71.94 60.90 132.8	17:30 marks	20 0.00 0.00 0.00	2	71.941. 60.909. 132.850.
1/1-12/2	11/88 15/89 15/19	Lee De Lano Lee De Lano Lee De Lano Greg Dias	(@ 100t) (@ 50t)	Tarres Tarres	132,8 132,8 132,8 132,8	17.30 mg/m 19.75 50.85 	0.00 0.00 0.00 0.00	2	
1/1-12/3 1/1-10/3 508TOT/ 10/16-11 1/1-5/3	11/88 15/89 12/31/89	Lee De Lano Lee De Lano Lee De Lano Greg Dias	(@ 100¢ (@ 50¢)		20.3 132.8 20.0 20.0 20.0 20.0 20.0 20.0 20.0 2	17.46 0 0	20 0.00 0.00 0.00 0.00 0.00	2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	71.941. 60.909. 132.850. 20.349. 43.237.
1/1-12/7 1/1-10/7 5USTOT: 10/16-11 1/1-5/3 SUBTOT:	11/88 15/49 2/31/69 1/90	Lee De Lano Lee De Lano Lee De Lano Greg Dias Greg Dias	(@ 100t (@ 50t)		132,8 132,8 132,8 132,8	17.10 mm m	0.00 0.00 0.00 0.00 0.00 0.00	0.00 2	71.941. 60.909. 132,850. 20,349. 43,237. 63,587.
1/1-12/3 1/1-10/3 508TOT/ 10/16-13 1/1-5/3 508TOT/ 5/29-7/ 8/1-12/	11/88 13/19 1/21/89 1/20 11/20 15/20	Lee De Lano Lee De Lano Lee De Lano Greg Dias Greg Dias Allen Short	(@ 100t (@ 50t)		20.30 132.8 20.00	17.16 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.00 0.00 0.00 0.00 0.00 0.00	0 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	71,941. 60,909. 132,850. 20,349. 43,237. 63,587.
1/1-12/7 1/1-10/7 5/08TOT/ 10/16-11 1/1-5/3 SUBTOT/ 5/29-7/ E/1-12/	11/88 15/89 1/90 AL 11/90 15/90	Lee De Lano Lee De Lano Lee De Lano Greg Dias Greg Dias Greg Dias Allen Short Allen Short	(G 1000 (G 1000 (G 200) (G 200)		20,3 20,4 20,5	49.55 37.91 2 49.33 34.33	0.00 0.00 0.00 0.00 0.00 0.00		71,941. 60,909. 102,850. 20,349. 43,237. 63,587. 24,614. 50,134. 77,748.
1/1-12/7 1/1-10/7 5UBTOT/ 10/16-11 1/1-5/3 SUBTOT/ 5/29-7/ E/1-12/	11/88 15/89 1/90 AL 11/90 15/90	Lee De Lano Lee De Lano Lee De Lano Greg Dias Greg Dias Allen Short	(G 1000 (G 1000 (G 200) (G 200)		132.8 132.8	17.10 more 29.75 50.85 49.55 37.91 % 24.31 34.33	0.00 0.00 0.00 0.00 0.00 0.00	Control Control	71,941. 60,909. 102,850. 20,349. 43,237. 63,587. 24,614. 50,134. 77,748.
1/1-12/7 1/1-10/7 2/1-10/7 10/16-17 1/1-5/3 SUBTOT 5/29-7/ E/1-12/	11/88 15/89 1/90 AL 11/90 15/90	Lee De Lano Lee De Lano Lee De Lano Greg Dias Greg Dias Allen Short Allen Short Allen Short	(@ 50%) (@ 50%) (@ 100% (@ 100%		20.3 20.4 20.4	17.10 more 29.75 50.85 49.55 37.91 2 27.46 0 0 24.33 24.33 24.33	0.00 0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	71.941.60.909.112.850.20,349.43.237.63.587.24.614.50.134.77.748.

J, Vickin Chring, Seam cary of the Market of Cirretters of Cirretters of Community Morestry and Carretters of Community Carretters and Community Carretters and Community Carretters of Carretters of Market of Carretters of Market of Cirretters of Market of Carretters of Market of Cirretters of Market of Cirretters of Carretters of Market of Cirretters of Carretters of Market of Carretters of Market of Carretters of Carr

నిడి. కెట్టించింది ఇక్కార్ స్ట్రిస్తున్న కొన్ని సిన్ని ప్రాథాలు అన్ని సిన్ని ప్రాథాలు కొన్నారు. అన్ని మన్స్ కోడింది కొడ్డి కొడ్డి కొన్ని కి.మీ. and the second

RESOLUTION NO. 91-32

APPROVING MEMORANDUM OF UNDERSTANDING AMONG MODESTO IRRIGATION DISTRICT, CITY OF MODESTO AND DEL ESTE WATER COMPANY FOR THE DOMESTIC WATER PROJECT

WHEREAS, the Modesto Irrigation District "MID", City of Modesto "COM" and Del Este Water Company "DEWC" have worked together to develop a Memorandum of Understanding to govern their relationship with regard to the design, finance, construction, operation and maintenance of the surface water treatment plant and distribution system facilities; and

WHEREAS, the "MID", "COM" and "DEWC" have agreed to the final Memorandum of Understanding as presented at today's meeting to this Board of Directors for consideration.

BE IT RESOLVED, That the Board of Directors of the Modesto Irrigation District does hereby authorize and direct the President and Secretary to execute the Memorandum of Understanding among Modesto Irrigation District, City of Modesto and Del Este Water Company. Reference is hereby made to said Memorandum of Understanding, filed as Document No. 4269 in the office of the District's Secretary, for full particulars.

Moved by Director Lyons, seconded by Director Cowan, that the foregoing resolution be adopted. Constant and all the constants

Upon roll call the following vote was had:

1 9724

Ayes: Directors Beck, Cowan, Billington, Kidd, Lyons Noes: Directors None

Absent: Directors None

The President declared the resolution adopted.

000

I, Vickie Ehrler, Secretary of the Board of Directors of the MODESTO IRRIGATION DISTRICT, do hereby CERTIFY that the foregoing is a full, true, and correct copy of a resolution duly adopted at a special meeting of said Board of Directors held on the 26th day of February, 1991. Lubu Calas

Secretary of the Board of Directors of the Modesto Irrigation District

באאפן שנשגני, גוננ ספודני

*1 .

MODESTO CITY COUNCIL RESOLUTION NO. 91-126

A RESOLUTION APPROVING A MEMORANDUM OF UNDERSTANDING WITH MODESTO IRRIGATION DISTRICT AND DEL ESTE WATER COMPANY FOR DESIGN, FINANCING AND CONSTRUCTION OF A SURFACE WATER TREATMENT PLANT, TRANSMISSION PIPELINES AND RESERVOIRS

BE IT HEREBY RESOLVED by the Council of the City of Modesto that the Memorandum of Understanding with Modesto Irrigation District and Del Este Water Company for design, financing and construction of a surface water treatment plant, transmission pipelines and reservoirs be, and it is hereby approved.

BE IT FURTHER RESOLVED that the execution of said Memorandum of Understanding by the designated city officials be authorized.

The foregoing resolution was introduced at a regular meeting of the Council of the City of Modesto held on the 26th day of February, 1991, by Councilmember Patterson, who moved its adoption, which motion being duly seconded by Councilmember Muratore, was upon roll call carried and the resolution adopted by the following vote:

Councilmembers: AYES:

Bird, Dobbs, Irizarry, Lang, Muratore, .

Patterson, Mayor Whiteside

NOES:

Councilmembers:

None

ABSENT: Councilmembers:

None

Original Signed By ATTEST:

NORRINE COYLE, City Clerk

The foregoing is a correct copy of the arisinal on this in this office which has met them muched and is now in full force

ALL OF SILV SLAM OF THE STY County of Stanislaus State of 10.75

تنسبر الدام،

(END OF APPENDIX A)